TECHNOLOGY TRANSFER MODEL FOR THE KLANG VALLEY MASS RAPID TRANSIT DEVELOPMENT PROJECT

ABDUL RAHMAN BIN HAMDAN

UNIVERSITI TEKNOLOGI MALAYSIA

TECHNOLOGY TRANSFER MODEL FOR THE KLANG VALLEY MASS RAPID TRANSIT DEVELOPMENT PROJECT

ABDUL RAHMAN BIN HAMDAN

A thesis submitted in fulfilment of the requirements for the award of the degree of Doctor of Philosophy

Razak Faculty of Technology and Informatics Universiti Teknologi Malaysia

OCTOBER 2021

DEDICATION

This thesis is dedicated to my parents, who taught me that the best kind of knowledge to have is that which is learned for its own sake. It is also dedicated to my wife, who taught me that even the largest task can be accomplished if it is done one step at a time. The thesis is also dedicated to my children who gave me the strength and patience that is needed to complete the journey.

ACKNOWLEDGEMENT

In preparing this thesis, I was in contact with many people: researchers, academicians and practitioners. They have contributed towards my understanding and thoughts process. In particular, I wish to express my sincere appreciation to my main thesis supervisor, Professor Madya Dr. Mohamad Syazli Fathi, for encouragement, guidance, criticism and friendship. I am also very thankful to my co-supervisor Professor Emeritus Dato' Ir. Ts. Dr. Zainai Mohamed for his wisdom, advice and motivation. Without their continued support and interest, this thesis would not have been the same as presented here.

I am also indebted to Universiti Teknologi Malaysia (UTM), the key staff of the KVMRT project procurement agency for their assistance and cooperation in supplying the relevant literature, resources and participation for completing the study.

My fellow postgraduate students should also be recognised for their support. My sincere appreciation also extends to all my colleagues and others who have provided assistance at various occasions. Their views and tips are useful indeed. Unfortunately, it is not possible to list all of them in this limited space. I am grateful to all my family members.

ABSTRACT

Realising the strategic leverage that large-scale public procurement has on local technology development, several countries introduced a Technology Transfer (TT) programme into their public procurement as part of capability-building. Recognising this leverage, the government of Malaysia has also introduced this Technology Transfer programme into some government strategic procurements to develop local industries. Malaysia still depends heavily on foreign technology in certain sectors and this over-reliance is due to the ineffectiveness of the TT model currently being implemented. Arguably, effective implementation of TT by the government would help Malaysia to reduce this dependency. However, there is a lack of studies measuring the effectiveness of TT in Malaysia's public mega-projects. This study provides an analysis of a case study involving the TT model used in a public rail infrastructure project in Malaysia. It aims to identify factors impacting the current effectiveness of the technology transfer process in the nation's mega infrastructure project and propose a framework. The study was conducted using a mixed-method analysis of qualitative and quantitative data. The Klang Valley Mass Rapid Transit (KVMRT) project was selected as a suitable case study due to the massive amounts of capital and technology needed for its completion. The final output of the study is a four-stage TT model for the KVMRT project. The first stage is a literature review in which TT models from several studies were reviewed. Based on this review and comparison analysis, a conceptual KVMRT TT model was developed. To further investigate the key factors of the TT process in the project, two stages of study: primary study 1 and primary study 2 were conducted. Primary study 1 was conducted qualitatively to test and evaluate the conceptual model developed during the early stage with the identified key experts. Primary study 2 was conducted quantitatively using factor analysis and Structural Equation Modelling (SEM) to test and evaluate further the findings from primary study 1. A total of 202 respondents were selected from personnel who were directly involved in the KVMRT project and TT programme to participate in a survey. The fourth stage was the model validation stage, where a focus group and case studies are used to validate the findings from the primary study 2. Based on these findings and the validation exercise, a final KVMRT-TT model was developed. Six main factors that are highly influential were identified in determining the effective and successful outcome of the TT programme in the KVMRT project, namely: "Technology Transfer Planning", "Transfer Environment", "Learning Environment", "Technology Provider's Characteristics", "Recipient's Characteristics" and "Technology Transfer Outcome". Among these factors, the "Learning Environment" was found to be the strongest factor influencing the TT outcome. This study made several recommendations to improve the TT process which include: (1) Improvement of current government policy on Technology Transfer, (2) Focus attention on innovations as one of the main criteria for a TT outcome, (3) Digitalization of the TT Process, (4) Proper measurement and evaluation of TT outcomes, and (5) Establishment of a Technology Transfer Office (TTO) in a government strategic procurement project.

ABSTRAK

Menyedari besarnya pengaruh perolehan kerajaan yang strategik dan berskala besar terhadap pembangunan teknologi tempatan, beberapa negara telah memperkenalkan Pemindahan Teknologi (TT) dalam perolehan awam mereka sebagai sebahagian peningkatan keupayaan. Kerajaan Malaysia dalam memperakui pengaruh ini juga telah memperkenalkan program Pemindahan Teknologi di dalam perolehan strategik kerajaan untuk membangunkan industri tempatan. Malaysia masih sangat bergantung kepada teknologi luar dalam sektor tertentu dan kebergantungan ini disebabkan oleh model TT semasa yang dilaksanakan tidak efektif. Oleh itu, pelaksanaan TT yang berkesan oleh kerajaan akan membantu Malaysia mengurangkan kebergantungan ini. Namun begitu, tiada kajian telah dibuat yang mengukur keberkesanan program Pemindahan Teknologi (TT) di dalam projek mega kerajaan ini. Kajian ini menyediakan analisis berdasarkan satu kes kajian model TT di dalam projek infrastruktur rel di Malaysia. Kajian ini adalah untuk mengenal pasti faktor yang memberi impak kepada keberkesanan semasa proses pemindahan teknologi dalam projek infrastruktur negara dan mencadangkan satu kerangka kerja. Kajian ini menggunakan kaedah analisis bagi data kualitatif dan kuantitatif. Projek Transit Aliran Deras Lembah Kelang (KVMRT) dipilih sebagai kes kajian kerana projek tersebut memerlukan dana awam yang besar dan juga keperluan teknologi untuk menyiapkannya. Hasil akhir kajian ini adalah empat peringkat TT untuk projek KVMRT berkenaan. Peringkat pertama adalah kajian literatur di mana model TT daripada beberapa kajian dikaji. Berdasarkan kajian dan analisis perbandingan, sebuah konsep model TT bagi KVMRT dibangunkan. Bagi menyiasat faktor-faktor penting proses TT di dalam projek, dua peringkat kajian: kajian utama 1 dan kajian utama 2 telah dilakukan. Kajian utama 1 dilakukan secara kualitatif bagi menguji dan menilai model konsep yang telah dihasilkan semasa peringkat awal kajian dengan pakar-pakar yang telah dikenal pasti. Kajian utama 2 pula telah dijalankan secara kuantitatif menggunakan analisis faktor dan Structural Equation Modelling (SEM) bagi menguji dan menilai penemuan kajian daripada kajian utama 1. Sebanyak 202 responden telah dipilih daripada kakitangan yang terlibat secara langsung dalam projek KVMRT dan program TT untuk turut serta di dalam soal selidik. Peringkat keempat adalah peringkat pengesahan dengan kumpulan fokus dan juga kajian kes digunakan untuk mengesahkan penemuan daripada kajian utama 2. Daripada penemuan dan pengesahan tersebut, sebuah model akhir KVMRT-TT telah dibangunkan. Terdapat enam faktor penting yang mempengaruhi keberkesanan dan kejayaan hasil program TT di dalam projek KVMRT iaitu: "Perancangan Pemindahan Teknologi", "Persekitaran Pemindahan Teknologi", "Persekitaran Pembelajaran", "Ciri-ciri Pembekal Teknologi", "Ciri-ciri Penerima Teknologi" dan "Hasil Pemindahan Teknologi". Di antara faktor-faktor ini, didapati "Persekitaran Pembelajaran" adalah faktor utama yang mempengaruhi hasil TT. Kajian ini mengetengahkan cadangan-cadangan bagi penambahbaikan proses TT termasuklah (1) Penambahbaikan polisi kerajaan bagi Pemindahan Teknologi, (2) Inovasi sebagai fokus dan kriteria utama hasil TT, (3) Digitalisasi proses TT, (4) Pengukuran serta penilaian yang wajar ke atas hasil TT dan (5) Penubuhan sebuah Pusat Pemindahan Teknologi (TTO) di dalam sebuah projek perolehan strategik kerajaan.

TABLE OF CONTENTS

TITLE

DECLARATION	iii
DEDICATION	iv
ACKNOWLEDGEMENT	v
ABSTRACT	vi
ABSTRAK	vii
TABLE OF CONTENTS	viii
LIST OF TABLES	xii
LIST OF FIGURES	xiv
LIST OF ABBREVIATIONS	xvi
LIST OF SYMBOLS	xviii
LIST OF APPENDICES	xix
CHAPTER 1 INTRODUCTION	1

1.1	Research Background 1			
1.2	Problem Statement	6		
1.3	The Rationale for Developing the KVMRT Tech Transfer Model	nology 7		
1.4	Research Questions	11		
1.5	Research Objectives	12		
1.6	Scope of the Study 12			
1.7	Research Stages	15		
	1.7.1 Conceptual Model	15		
	1.7.2 Primary Study 1	16		
	1.7.3 Primary Study 2	16		
	1.7.4 Model Validation	17		
	1.7.5 Findings and Recommendations	17		
1.8	Thesis Structure	17		

CHAPTER 2 LITERATURE REVIEW			21	
2	2.1	Introduction		
2	2.2	Definition of Technology Transfer 22		
2	2.3	Technology Transfer Office (TTO)	24	
2	2.4	Evolution of a Technology Transfer Model	25	
		2.4.1 Appropriability Model	26	
		2.4.2 Dissemination Model	27	
		2.4.3 Knowledge Utilisation Model	28	
		2.4.4 Interrelationship Model	29	
		2.4.5 Role Shifting Model	30	
		2.4.6 Comparison of the Technology Transfer Models	31	
2	2.5	Technology Transfer Model Evolution facilitated by Malaysia's National's Policies	33	
2	2.6	Review of Selected Technology Transfer Models to develop the KVMRT Conceptual Technology Transfer Model42		
2	2.7	Comparison of Technology Transfer Models 53		
2	2.8	The Klang Valley Mass Rapid Transit Project (KVMRT)54		
2	2.9	The Offset / Industrial Collaboration Programme (ICP) Policy 55		
2	2.10	TT Factors and Sub-Factors 5		
2	2.11	Development of the Conceptual KVMRT Technology Transfer Model		
2	2.12	Chapter Summary 8		
CHAPTER	3	RESEARCH METHODOLOGY	87	
3	3.1	Introduction	87	
3	3.2	Research Design 88		
		3.2.1 Stage 1 – Conceptual Model Development	90	
		3.2.2 Stage 2 – Primary Study 1	90	
		3.2.3 Stage 3 – Primary Study 2	95	
		3.2.4 Stage 4 – Model Validation	104	
3	3.3 Justification for the Selection of the Research Methodology		110	
3	3.4	Chapter Summary 11		

CHAPTER 4	RESULTS AND ANALYSIS 115			
4.1	Introd	Introduction		
4.2	Primary Study 1		115	
	4.2.1	Key Outcomes from the Primary Study 1 Stage	116	
	4.2.2	Gap Analysis	117	
4.3	Prima	ry Study 2	122	
	4.3.1	Mean and Standard Deviation	123	
	4.3.2	Correlation	125	
	4.3.3	Analysis of Variance (ANOVA)	126	
	4.3.4	Factor Analysis for the Conceptual Model	127	
	4.3.5	Structural Equation Modeling (SEM)	131	
4.4	Valida	ation of Technology Transfer Model	141	
	4.4.1	Focus Group	141	
	4.4.2	Case Study Benchmarking	146	
4.5	Chapt	er Summary	170	
CHAPTER 5	FIND	INGS AND DISCUSSION	173	
5.1	Introd	uction	173	
5.2	Resear Exerci	rch Outcomes from the Primary Study and Validatio ise – Final KVMRT Technology Transfer Model	n 173	
	5.2.1	Most influential factor in the technology transferoutcome	er 175	
	5.2.2	Government's influence in the KVMRT Technolog Transfer Model	у 176	
5.3	Gap Analysis Findings and Recommendations 17		177	
5.4	Way F	Forward for the Technology Transfer Process	190	
	5.4.1	Improving the nation's technology transfer policy	190	
	5.4.2	Innovation as the primary driver of the technolog transfer process	y 191	
	5.4.3	The Digitalisation of the Technology Transfer Process	193	
	5.4.4	Proper Measurement and Evaluation for TT Outcomes	195	
	5.4.5	Introducing a Technology Transfer Office (TTO) in th KVMRT Technology Transfer Process	e 196	
5.5	Chapt	er Summary	196	

CHAPTER 6	CONCLUSION AND RECOMMENDATIONS	199
6.1	Introduction	199
6.2	Objective 1: To identify the current process flows in the Klang Valley Mass Rapid Transit (KVMRT) Technology Transfe Programme	g r 199
6.3	Objective 2: To determine the factors and sub-factors that contribute to the effectivess and success of the technology transfer in the KVMRT development project	ut y 201
6.4	Objective 3: To develop a Technology Transfer Model for the KVMRT project	e 202
6.5	Objective 4: To validate the Technology Transfer Model	204
6.6	Contributions to Knowledge	205
6.7	Limitations of the Research	208
6.8	Future Works	210
6.9	Concluding Remarks	211
REFERENCES		213
LIST OF PUBLI	LIST OF PUBLICATIONS 27	

LIST OF TABLES

TABLE NO.	TITLE	PAGE
Table 1.1	Local content in major rail infrastructure projects in Malaysia	4
Table 2.1	Comparison of the Technology Transfer Models	31
Table 2.2	Summary of related national policies that facilitate technolo transfer	ogy 40
Table 2.3	Comparison summary between the Technology Transfer Models	51
Table 2.4	Summary of literature review on factors and sub-factors	57
Table 2.5	Comparison with similar studies on technology transfer in	20
Table 2.1	KVMR1 project	80
Table 2.2	Primary study research tools and analysis methods	95
Table 3.2 Table 3.3	Stages of the study, research objectives, research methods a output	99 and 111
Table 4.1	Summary of the current KVMRT technology transfer process a weaknesses	and 119
Table 4.2	Mean and Standard Deviation $(N = 202)$	124
Table 4.3	Summary of Factor Extraction using EFA analysis	130
Table 4.4	Model Fit Indices for the SEM Measurement Model	133
Table 4.5	Summary results of CFA	135
Table 4.6	Discriminant Validity	137
Table 4.7	Model Fit Indices for the SEM Structural Model	140
Table 4.8	Summary of the key points from the focus group	142
Table 4.9	Case study unit of analysis and description	147
Table 4.10	Technology transfer project milestones and timeframe (Case Stu 1)	ıdy 150
Table 4.11	Key findings from Case Study 1	151
Table 4.12	Technology transfer project milestones and timeframe (Case Stu 2)	ıdy 153
Table 4.13	Key findings from Case Study 2	155

Table 4.14	Technology transfer milestones and timeframe (Case Study 3)	158
Table 4.15	Key Findings from Case Study 3	159
Table 4.16	Technology transfer milestones and timeframe (Case Study 4)	161
Table 4.17	Key Findings from Case Study 4	162
Table 4.18	Technology transfer milestones and timeframe (Case Study 5)	164
Table 4.19	Key Findings from Case Study 5	165
Table 4.20	Summary of key factors from the case studies	166
Table 4.21	Technology transfer factors and sub-factors for evaluation by the WPC	ne 169
Table 6.1	Proposed action and output for the current KVMRT technolog transfer process and the identified weaknesses	y 200

LIST OF FIGURES

FIGURE NO	. TITLE	PAGE	
Figure 1.1	Malaysia's railway development timeline (1885 – 2020)	8	
Figure 1.2	Scope of this Study		
Figure 1.3	Research stages	15	
Figure 2.1	Example of an Appropriability Model	26	
Figure 2.2	Example of Dissemination Model	27	
Figure 2.3	Example of Knowledge Utilisation Model	28	
Figure 2.4	Example of Interrelationship Model	29	
Figure 2.5	Example of Role Shifting Model	30	
Figure 2.6	Conceptual KVMRT Technology Transfer Model	83	
Figure 3.1	Stages of the Study	87	
Figure 3.2	Triangulation method	89	
Figure 3.3	SEM research methodology process flow	96	
Figure 3.4	Participants' demographic breakdown (By type)		
Figure 3.5	Participants' demographic breakdown (By age)		
Figure 3.6	Participants' demographic breakdown (By experience involvement in a technology transfer programme)	of 101	
Figure 3.7	Participants' demographic breakdown (By experience participating in a Technology Transfer Programme)	of 102	
Figure 3.8	Sequence of activities for case studies approach	109	
Figure 4.1	Gap Analysis framework and process flow for KVMRT		
	TT programme	118	
Figure 4.2	KVMRT technology transfer conceptual model	128	
Figure 4.3	The SEM measurement model	132	
Figure 4.4	Final SEM structural model	139	
Figure 5.1	Final KVMRT Technology Transfer Model	174	
Figure 5.2	Evolution of Technology Transfer Models in relation to nation policy	nal 193	
Figure 6.1	KVMRT Technology Transfer Model based on Structural		

	Equation Modelling (SEM)	202
Figure 6.2	Validated final KVMRT Technology Transfer Model	204

LIST OF ABBREVIATIONS

AC	-	Alternating Current
ANOVA	-	Analysis of Variance
AVE	-	Average Variance Extracted
BIM	-	Building Information Modelling
CFA	-	Confirmatory Factor Analysis
CR	-	Composite Reliability
DC	-	Direct Current
EFA	-	Exploratory Factor Analysis
EIA	-	Environmental Impact Assessment
EOI	-	Export-Oriented Industrialization
ERL	-	Express Rail Link
FAT	-	Factory Acceptance Test
FTZ	-	Free Trade Zone
GIS	-	Gas-insulated Switchgear
GNI	-	Gross National Income
HV	-	High Voltage
ICP	-	Industrial Collaboration Programme
ICT	-	Information and Communication Technology
IMP	-	Industrial Master Plan
IMU	-	ICP Management Unit
IRD	-	ICP Requirement Document
ISD	-	ICP Strategic Document
ISI	-	Import-Substituting Industrialization
KLIA	-	Kuala Lumpur International Airport
KM	-	Kilometres
КМО	-	Kaiser-Meyer-Olkin
KTMB	-	Keretapi Tanah Melayu Berhad
kV	-	Kilovolt
KVMRT	-	Klang Valley Mass Rapid Transit

LRT	-	Light Rail Transit	
MIGHT	-	Malaysian Industry Government Group for High Technology	
MITI	-	Ministry of International Trade and Industry	
MNCs	-	Multi-National Companies	
MOF	-	Ministry of Finance	
MRT	-	Mass Rapid Transit	
OECD	-	Organization for Economic Cooperation and Development	
OCS	-	Overhead Catenary System	
OEM	-	Original Equipment Manufacturer	
OJT	-	On-Job-Training	
ORD	-	Offset Requirement Document	
OSA	-	Office of Science Advisor	
PDP	-	Project Delivery Partner	
PEMANDU	-	Performance Management and Delivery Unit	
PERODUA	-	Perusahaan Otomobil Kedua	
PROTON	-	Perusahaan Otomobil Nasional	
QA	-	Quality Assurance	
QC	-	Quality Control	
R&D	-	Research and Development	
SBK	-	Sungai Buloh-Kajang	
SCADA	-	Supervisory Control and Data Acquisition	
SEM	-	Structural Equation Modelling	
SF6	-	Sulphur Hexafluoride	
SSP	-	Sungai Buloh-Serdang-Putrajaya	
S&T	-	Science and Technology	
TDA	-	Technology Depository Agency	
TIO	-	Technology Innovation Office	
TT	-	Technology Transfer	
TTO	-	Technology Transfer Office	
UTC	-	University Technology Centre	
UTM	-	Universiti Teknologi Malaysia	
WPCs	-	Works Package Contractors	

LIST OF SYMBOLS

α	-	Cronbach Alpha
x	-	Mean
r	-	Pearson Correlation Coefficient
σ	-	Standard Deviation

LIST OF APPENDICES

APPENDIX	TITLE	PAGE
Appendix A	Question Form (Primary Study 1)	231
Appendix B	Transcribed Interviews (Primary Study 1)	237
Appendix C	TT Models from the Literature Review	251
Appendix D	External Expert Verification of Question Interviews	257
Appendix E	Survey Form (Primary Study)	259
Appendix F	Theme Frequencies from NVivo 12 Analysis	264
Appendix G	Pearson Correlation Matrix	268

CHAPTER 1

INTRODUCTION

1.1 Research Background

Leveraging public procurement as a strategic tool for technology development and innovation is not considered as something new (Maleki *et al.*, 2017; Uyarra and Flanagan, 2009). Many countries around the world have utilised their public procurement as a strategic tool for technology transfer and capability-building through mechanisms such as Offset and Counter-trade programmes (Balakrishnan, 2007; Hamdan, 2015). However, studies of the effectiveness of utilising a government's procurement policy to effectively develop local industrial capability, especially at the level of companies and firms, are still lacking (Edler *et al.*, 2013). Therefore, this study aims to develop a technology transfer model in order to study the effectiveness of the technology transfer process in a public infrastructure project based on the government's technology transfer policy.

The technology transfer process itself is not something considered to be new. Technology transfer was already happening during the Neolithic age, where human civilisation was starting to develop new knowledge and technology and transforming themselves from nomadic hunters and gatherers to settled farmers by learning new knowledge and processes (Segman, 1989; Thohari *et al.*, 2013). This new knowledge and technology were often shared with others as ways to create linkages and contact between civilisations. Thus, technology transfer has occurred throughout all civilisations worldwide. For example, Islamic civilisation, as the global leader in scientific knowledge to the western world through several territories such as Al-Andalus and Sicily during the early middle ages (Segman, 1989; Tahri, 2012). This historical fact shows that technology and knowledge transfer occur naturally whenever two civilisations meet. History also shows that technology transfer can trigger an

industrial revolution, such as the American industrial revolution during the 19th century. The industrial revolution started from the technology transfer of British textile expertise, technical assistance and operating personnel to America that resulted in faster textile production (Cameron, 1960; Irwin and Moore, 1991; Mohajan, 2019). Therefore, technology transfer has played an important part in shaping the history of the world and the development of countries throughout history. History shows the critical role of technology transfer in the transformation of a country such as Japan after World War Two. The result of a technology and knowledge transfer in the area of quality management from Edwards Deming in the United States to Japan led to Japan's economic revolution (Mandel, 2012). Edwards Deming pioneered quality control techniques in manufacturing and shared them with Japanese automotive manufacturers. The Japanese manufacturers adopted Deming's quality control techniques and later created their own quality control methods (Mandel, 2012). The Japanese manufacturers were able to utilize the knowledge received from Deming and customize it according to their needs and environment. The technology transfer between Deming and the Japanese automotive manufacturers shows that with the right factors and sub-factors in place, the process can lead to a significant outcome that would benefit the economy of a country tremendously. Japanese manufacturers since then have further improved their manufacturing and quality management methods to become a reference for the world.

In the Malaysian local context, technology transfer has always been used by the government as leverage to develop local industries and to transform Malaysia from a developing country to fully developed nation status. The mechanism used by the government to facilitate technology transfer has always been in the form of national policies and guidelines (MOF, 2014). The national policies serve as the governing mechanism for the industries and academia in implementing technology transfer for the benefit of local industries. The government of Malaysia has always used construction projects as a basis for importing technology through international technology transfer (Omar *et al.*, 2008). The construction projects that have attracted top foreign companies to participate in the project have allowed the local construction companies to participate and learn the foreign companies' construction techniques and process. Example of these large infrastructure mega-projects from which the local construction industry has previously benefitted are the Petronas Twin Towers and Kuala Lumpur International Airport or KLIA (Omar *et al.*, 2010; Omar *et al.*, 2008). The SMART Tunnel construction project in Kuala Lumpur, Malaysia for example has enabled technology transfer from foreign technology to the local industry in adapting a new tunnelling technology methods by leveraging the SMART Tunnel construction (Isah and Ali, 2015).

Technology transfer has also been prominent in other key sectors such as the rail support industry. Due to the Government's effort in modernising the local rail infrastructure in the 1990s, the rail support industry has been the beneficiary of many different rail technologies (MIGHT, 2014a). However, the local railway industry in Malaysia is currently very much dependent on international Original Equipment Manufacturers (OEM), especially in railway systems technology. Only limited amounts of railway systems technology have been produced by local companies, possibly due to the ineffective technology transfer process being implemented (MIGHT, 2014a). At the time the study was conducted, Malaysia's local capability in terms of rail infrastructure systems such as Electric Trains, Signalling Systems, Maintenance, Repair and Overhaul was still critically lacking (MIGHT, 2014a). The study by MIGHT identified the technology gaps for the local railway system. Effective measures by the government are still needed to ensure that this critical technology gap can be closed by the local rail companies in meeting both local and global demands.

A previous study on the technology transfer during Malaysia's Express Rail Link (ERL) project was also conducted (Mohamed *et al.*, 2015). Based on this study, there is already a pocket of local workers who are skilled and knowledgeable in building modern railway infrastructure here in Malaysia (Mohamed *et al.*, 2015). This scenario is due to their involvement in a previous government's rail infrastructure projects such as PUTRA LRT, STAR LRT and the KTM Komuter project. However, the study stated that the pool of skilled workers is most experienced in Light Rapid Transit systems. Therefore, when the government introduced a high-speed rail system, such as the ERL project, local industry was facing significant gaps in technology, knowledge and skill (Mohamed *et al.*, 2015). Based on this, any new railway infrastructure project that uses new rail systems will need to acquire new knowledge and engage in technology transfer. As the government are now developing Malaysia's first and second Mass Rapid Transit rail networks, new rail systems will be introduced into the country accompanied by the need for technology transfer to the local industries to sustain and maintain the rail system in the long run.

Over the years, the government has spent a massive amount of public funds to modernise the rail infrastructure in Malaysia. According to the SPAD Master Plan, the government has invested more than RM50 billion in upgrading the rail systems in the country (SPAD, 2014). This modernisation includes the government's purchase of new electric trains and also rail supporting equipment and systems such as signalling and automated fare collection (Nee, 2013). There have also been several rail network extension projects that extend the current LRT line to cover more areas. However, with all the rail modernisation projects over the years, the local contribution to these projects is significantly low, especially in the rail systems (MIGHT, 2014a). As can be seen in Table 1.1, there is minimal local content, particularly in the rail systems for the three major rail infrastructure projects in the past ten years.

Rail Projects	Local Content (Percentage)		
	Rolling Stock	Systems	Infra (Civil
			Works)
Kelana Jaya Line	30%	0-19%	100%
Expansion			
Ampang Line	Nil	Nil	100%
Expansion			
MRT Sg Buloh to	30%	30%	100%
Kajang Line			

Table 1.1Local content in major rail infrastructure (Hamdan, 2015; MIGHT,2014a)

Table 1.1 shows that the local rail operator is still relying heavily upon foreign technology to build the railway infrastructure as can be seen in the LRT extension project, in particular the rail systems utilise less than 30% local content. However, for the KVMRT project, the government has realised the need to increase the local content. Therefore, the government have imposed a mandatory minimum of 30% local content on the project contractors.

The government has also imposed the Offset programme - now known as the Industrial Collaboration Programme (ICP) - requirement on the foreign contractors who have a contract value of more than RM100 million in the KVMRT project (MOF, 2014; MRTC, 2013). By having this mandatory obligation specified in the contract, the foreign contractors have no choice but to offer a proper and structured technology transfer programme to local firms that can be accepted by the government (Hamdan, 2015). In KVMRT line 1, six work packages triggered the Offset programme when awarded to a foreign company with a contract value of more than RM100 million. These packages are the Electric Trains, the Signalling and the Train Control System, the Power Supply and Distribution System, Track Work, Automated Fare Collection and Tunnelling Works (MRTC, 2013). The main objective of the KVMRT Offset programme is to develop Malaysia's local capability and capacity, especially in the areas of rail system technologies. However, based on the study by MIGHT, the local railway industry still heavily relied on foreign OEM's technology and there is still a lack of localization in recent rail infrastructure projects in the country (MIGHT, 2014a). This critical lack of localization, even with several big rail infrastructure project initiated by the government in the past year, shows that there is a need to study whether the technology transfer programme implemented by the works package contractors in the KVMRT project is effective in meeting its intended objectives so that the project will not repeat the failure of past technology transfer programmes in local public railway infrastructure projects.

The KVMRT project is one of the projects that has triggered the mandatory Offset requirement (Nee, 2013). From the tender requirement, most of the rail system works package contractors are required to deliver a technology transfer programme for the local companies to be monitored by the project procurement agency and the government (Nee, 2013). One of the main objectives of the KVMRT Offset programme is to enhance the local rail capability by leveraging a foreign contractor's technology and knowledge. Other objectives include attaching local companies as recipients of the technology transfer programme, the government only monitors the intended milestones and deliverables (MRTC, 2013). The soft aspect of the technology transfer programme, such as the critical success factors for an effective technology transfer process, is not being monitored closely. The preliminary assumptions are that

the parameters for the efficiency of the technology transfer will be multi-faceted and largely dependent on many factors, including the readiness of local recipients. There are also other interrelated factors in TT as described in previous studies that show that the TT process also depends on the absorptive capacity of the recipient and the preexisting technology gap (A. Bakar, 2006; Omar *et al.*, 2011).

1.2 Problem Statement

In the context of Malaysia's rail support industry, the nation still depends heavily on foreign technology to develop the local rail infrastructure (MIGHT, 2014a). This over-reliance on foreign rail technology is due to the ineffective technology transfer model currently being implemented by the government to develop the local rail support industry. Even though the government has introduced technology transfer programmes into public procurement, there is still a lack of localization especially in the rail support industry (MIGHT, 2014a). The lack of localization even with numerous public rail projects implemented by the government over the years shows the ineffectiveness of the current technology transfer programmes being implemented in public rail projects. Globally, technology development is increasing at a tremendous rate and therefore technologies are now becoming obsolete faster than before. As the global economy is entering the fourth industrial revolution, the technology disruption will become even more frequent in various industries and sectors. Therefore, the existing technology transfer model might no longer be relevant to support today's faster cycle of technology development. As Choi (2009) has highlighted, the technology transfer model now also needs to incorporate innovation as one of its primary outcomes (Choi, 2009). However, there are still gaps in terms of the connection between the performance of a company and the effectiveness of technology transfer. The gaps can be closed by identifying the factors and sub-factors that influence the technology transfer process from start to finish using a technology transfer model as reference and analysis. Therefore, the main research activity is to identify and measure the current effectiveness of the technology transfer process in the government's rail infrastructure project by studying the key factors and sub-factors affecting the effectiveness in a technology transfer programme by developing a technology transfer model. This Technology Transfer Model can be developed and analysed by identifying the factors and sub-factors related to the technology transfer process and its interrelationships. This model can be developed more accurately by using the quantitative approach where the key success factors and their relationship with other key factors can be identified and analysed. Based on the literature review of the national policies that facilitate technology transfer, there are no studies that have measured the effectiveness of this policy in the country. There is also a lack of studies analysing the effectiveness of technology transfer in a mega-project in Malaysia and how it impacts the local industries. Several studies, such as technology transfer in the ERL project and on the KVMRT project, were conducted based on qualitative methods (Mohamed et al., 2015; Omar et al., 2019). Also, the previous study on the ERL project and KVMRT project did not develop a technology transfer model that takes into account the transfer environment factor, e.g. whether government policy can have a direct or indirect influence on the technology transfer programme in the project. By having a developed KVMRT TT Model, the technology transfer process in the KVMRT project can be further analysed and improved. The effectiveness of the technology transfer process and the outcomes can be quantified by studying and validating a Technology Transfer Model. This can later be used for future public rail project technology transfer programmes as a means to further improve the technology transfer process that would eventually increase the localization of the rail support industry for the nation.

1.3 The Rationale for Developing the KVMRT Technology Transfer Model

In Malaysia, the rail support industry encompasses any rail-related activities such as research, design and development, infrastructure construction, maintenance, repair and overhaul and the final decommissioning of any rail-related types of machinery and vehicles (MIGHT, 2014a). Looking back at Malaysia's rail development timeline and history, as can be seen in Figure 1.1 below, the rail support industry in Malaysia was started exceedingly early even before the country's independence. The first railway track built in the country was in the year 1885 with a total length of 12.8km from Taiping to Port Weld. The British government entirely

financed the rail infrastructure and system at that time due to the need for transporting the mined tin to the port (Lowtan and Sussman, 2004).



Figure 1.1 Malaysia's railway development timeline (1885 -2020) (Lee and Sipalan, 2018; MIGHT, 2014a; SPAD, 2014)

However, once the country gained its independence, the government formed the "Keretapi Tanah Melayu" or KTM and later during the 1990s, transformed it into "Keretapi Tanah Melayu Berhad" or KTMB (Lowtan and Sussman, 2004). The modernisation of the rail infrastructure only started in 1995 with KTMB's first electrified track with the first electric train operating from Kuala Lumpur and Rawang (Mohamad, 2003). Malaysia also upgraded its rail infrastructure by implementing the Light Rail Transit (LRT) network in the early 1990s. The two LRT systems introduced were the STAR LRT and PUTRA LRT led by the private sector (Mohamad, 2003). In the effort to further upgrade the railway lines in the country, the electrified doubletracking project and also the LRT extension line project were implemented by the government from 2009 to 2011 (Mohamad, 2003). The government has also implemented several other metro lines such as the KVMRT Line 2 that runs from Sg Buloh to Putrajaya and also LRT 3. A new mega-rail-infrastructure development that has been in the planning and implementation mode is the High-Speed Rail Line connecting Kuala Lumpur to Singapore and also the East Coast Rail Line connecting Malaysia's west and east coasts (Lee and Sipalan, 2018; SPAD, 2014). This rail infrastructure project is likely to introduce modern rail system technology into the country on which the local industry can capitalise to grow their capability further.

As the population of the country was growing at a rapid pace, the government realised the need for more public railway infrastructure. According to PEMANDU, the railway infrastructure remains one of the critical components of the economic growth of the country (PEMANDU, 2012). However, the country's dependence on foreign rail technology is still high even though the rail support industry in Malaysia had been established before the country's independence. Malaysia's local rail operator, as well as rail integrator, depends heavily on foreign OEM technologies. As can be seen in the KVMRT Line 1 project, most of the rail system contractors and its technology are from foreign countries (MRTC, 2013). The local companies are still lacking in terms of technology and capability in rail systems such as rolling stock and signalling systems (MIGHT, 2014a). This dependence on foreign technology and systems has resulted in a massive outflow of capital from Malaysia to other countries. This outflow of money can be further reduced if the local industries are capable of supporting the rail infrastructure projects through localization of parts, components and skilled manpower.

The local industries' dependence on foreign technology can be reduced if the government puts in place a proper structure for technology transfer from the foreign OEMs to the local industries. Studies have shown that the local contractors can increase their technical capability by leveraging their working collaboration with the foreign OEMs (A. Bakar, 2006; Omar *et al.*, 2012). The efficiency of technology adoption can be increased if certain factors and sub-factors are introduced into the TT process (Oliveira and Teixeira, 2010; Waroonkun, 2007). Based on Malaysia's previous rail infrastructure projects such as the Express Rail Link (ERL), technology transfer opportunities can be capitalised, especially during the development of the rail infrastructure (Mohamed *et al.*, 2015). With proper planning and a formal structure, the technology that is being introduced by the foreign contractors can be absorbed by the local contractors (A. Khan, 2011; Waroonkun, 2007).

The Klang Valley Mass Rapid Transit Project (KVMRT) is selected as a suitable case study for this research because of the massive public capital used for procurement as well as the technology that is needed to complete the infrastructure development. The KVMRT Lines are also the first Mass Rapid Transit metro line project being developed and built in Malaysia. The previous metro rail project - the LRT project - falls under the Light Rail category. Due to these elements, the project can be considered as an appropriate platform to study the technology transfer process

and identify the critical success factors of the process during its planning and implementation phases.

The rationale for developing a technology transfer model for the KVMRT project arises from the lack of such a model for the rail support industry (Dinakar, 2011). Currently, there is no technology transfer model being developed using the quantitative approach in the Malaysian context. There has also been a lack of study in terms of identifying technology transfer factors related to the development of a national rail infrastructure. The literature review of technology transfer models shows that most of the previous studies used a qualitative approach in developing the technology transfer model. Therefore, there is a need to develop a TT model that is based on quantitative data and analysis. With substantial public funds and capital being pumped by the government into mega-construction projects such as KLIA, LRT and KVMRT infrastructure projects, there is a need to study their effectiveness in promoting technology transfer to local industries. There have been technology transfer programmes in other related rail support industry projects in Malaysia, such as LRT and ERL (Mohamed et al., 2015). However, there is a lack of effort to develop a technology transfer model based on a national public rail infrastructure project, especially using quantitative methods and analysis. As more public rail projects begin, a "Technology Transfer Model" for public rail infrastructure projects will benefit the country by improving the technology transfer process and at the same time helping to increase the local rail support industry's capability and knowledge.

Another rationale for selecting the Klang Valley Mass Rapid Transit infrastructure development project as the basis for the development of the technology transfer model is due to the massive public funds being used in its implementation. The KVMRT project is one of the biggest mega-structures ever financed by the government of Malaysia with the Sungai Buloh-Serdang-Putrajaya (SSP) Line budgeted at under RM40 billion (Kaur, 2016). The nature and scope of this large modern rail infrastructure development project will usually attract top players to participate, as can be seen in public rail projects in China (Chan and Aldhaban, 2009). Due to this factor, the KVMRT project serves as an excellent platform for datagathering using statistical methods due to most of the works package contractors participating in the project being established contractors both domestic and foreign (Nee, 2013). Foreign contractors participating in the project will bring their internal methods and technologies into the project and therefore will create high value-added activities with which local industries can collaborate and learn (Nee, 2013).

A valid reason for developing the KVMRT technology transfer model is to see the effectiveness of the government's policy in facilitating and governing a technology transfer programme in a public-funded project. The effectiveness of the TT model can be determined using quantitative analysis as previous studies on TT Models have done (Purushotham et al., 2015; Waroonkun, 2007). The existing national policies, such as the ICP policy, have served as the guideline for the procurement agency to implement technology transfer since 2013 (TDA, 2016). Improvements have been made to the policy since then with several amendments being made to the treasury circular in recent years. The improvements to the policy only focus on the mechanism of the ICP monitoring and not on improvements to the effectiveness of the technology transfer process flow (Hamdan, 2015; MOF, 2014). There is also a lack of studies to measure the effectiveness of this policy in ensuring that technology transfer is being implemented successfully in a public project, especially using quantitative analysis. The development of the KVMRT TT model will serve as a future reference for any procurement agencies or government-related bodies to plan and coordinate a TT programme that will effectively meet its intended objectives.

1.4 Research Questions

The research questions of the study that served as the guide for the entire research are as follows: -

- 1. What are the current process flows of the KVMRT Technology Transfer programme?
- 2. What are the underlying factors and sub-factors that influence the effectiveness of the KVMRT Technology Transfer process and make it a successful TT programme?

3. What is the validated Technology Transfer Model that can be used for the KVMRT Technology Transfer Process flow and for similar future TT Programmes?

1.5 Research Objectives

The main objectives of the study are as follows: -

- To identify the current process flows in the Klang Valley Mass Rapid Transit (KVMRT) Technology Transfer programme,
- 2. To determine the factors and sub-factors that contribute to the effectiveness and success of the technology transfer in the KVMRT development project,
- 3. To develop a Technology Transfer Model for the KVMRT project,
- 4. To validate this Technology Transfer Model.

1.6 Scope of the Study

In developing the technology transfer model, the study focuses on the Klang Valley Mass Rapid Transit development project as a case study. Line 1 of the KVMRT line was completed in 2017 (Ahmad and Sivanandam, 2017) and a further line is under construction with expected completion in 2022 (Kaur, 2016). As can be seen in Figure 1.2 below, the KVMRT project timeline started from the railway scheme design up until the line completion. The entire duration of the project took about eight to ten years. Based on the limitation and scope of this study, it will only focus on the technology transfer that occurs during the project key implementation stage which consists of contract procurement and the construction phase of the KVMRT Line 1 and the Line 2 development phase. The justification for selecting this timeline period is due to the foreign and local contractors' active participation and also the high volume of works and cost involved during the procurement and construction stages (Burmistrov *et al.*, 2017). The involvement and participation of these local and foreign

contractors during these stages will normally allow technology transfer to happen. Therefore, the main target respondents for the all the study stages are the personnel who directly participated in the KVMRT technology transfer programme either as participants or key personnel who manage and monitor the technology transfer process. However, for the reliability of the study, the respondents stated they have zero knowledge of technology transfer but have previously participated in a technology transfer programme in the questionnaire during primary study 2 would be accepted as a valid respondent to the study. The reason that this filtering question is in place in the questionnaire is to ensure only respondents actually have experience in technology transfer would be able to provide reliable feedback.

The procurement stage is an important area to study in terms of the preplanning of a technology transfer programme before the construction stage starts. The construction stage is where the implementation and integration of the rail system occurs and where many of the rail systems are provided by foreign suppliers (Nee, 2013). Therefore, the construction stage offers a suitable platform and case study for developing the technology transfer model. However, the evidence whether technology transfer has occurred during the construction stage will also be examined and validated during the initial study stage. The targeted respondents consist of the staff from the civil contractors, sub-contractors, rail system contractors and consultants as well as the project owner and the project delivery partner for the KVMRT project that either participated in the KVMRT technology transfer programme or managed and monitored the technology transfer process. Due to the extensive nature of this project, most of the civil contractors involved are among the largest in Malaysia. Most of the rail system contractors involved in this project are linked to established global companies that have extensive experience in rail infrastructure projects (MRTC, 2013).



Figure 1.2 Scope of this Study

The selection of the KVMRT project as the case study for this research is also due to the government's mandatory requirement for a technology transfer programme stated in the tenders for the KVMRT project. The binding obligation imposed by the government provides an excellent platform for studying a technology transfer programme because all the selected contractors involved will have to establish one. However, at the same time, the obligation may create an artificial and closed technology transfer environment that is not driven by market needs. Therefore, a limitation of the study is the omission of market needs as an influencing factor. The developed Technology Transfer Model, however, should be able to enhance the effectiveness of technology transfer programmes in similar government development projects (Abdul Wahab *et al.*, 2012a; Waroonkun, 2007).

1.7 Research Stages

In order to achieve the research objectives outlined, the research methodology includes qualitative and quantitative methods, where each step is planned and conducted to achieve the study's objectives. As shown in Figure 1.3 below, the development of the KVMRT Technology Transfer Model consists of five milestones. The milestones are the development of the conceptual model, primary study 1, primary study 2, model validation and, finally, the findings and recommendations. Chapter 3 provides a detailed explanation of the research methods used for achieving the targeted milestones in all stages of the study.



Figure 1.3 Research stages

1.7.1 Conceptual Model

In developing the conceptual model, an extensive literature review has been conducted to benchmark and analyse existing technology transfer models from previous studies. The study identifies the factors and sub-factors of the technology transfer process and their inter-relationships. Based on comparison analysis, a selected technology transfer model provided a suitable benchmarked TT model for the study. The detail of the technology transfer models and the development of the conceptual model are described in Chapter 2. The conceptual model serves as the basis for the further identification of factors and the relationships between them in later stages of this study.

1.7.2 Primary Study 1

Primary study 1 was conducted to test and evaluate the conceptual model developed during the early stage with the identified key experts. The key experts are filtered and selected based on specific criteria. One of the main criteria for selection as a respondent is that the personnel should be directly involved in managing or monitoring the KVMRT technology transfer programme. The filtering criteria ensure that primary study 1, which uses qualitative methods and tools, will receive reliable and high-quality feedback. The qualitative approach provides the study with potential new factors and sub-factors gathered from the key experts. The analysis and findings from primary study 1 are discussed in detail in Chapter 4.

1.7.3 Primary Study 2

Primary study 2 was conducted to test and evaluate further the findings from the primary study 1. The findings of the technology transfer factors and the relationship between the factors discovered during primary study 1 were then analysed quantitatively during the primary study 2 stage. Primary study 2 uses a questionnaire approach to gather feedback and data. About 435 questionnaires were distributed to participants in the technology and knowledge transfer programme in the KVMRT project. About 306 survey forms were returned of which 104 were rejected. Finally, the data provided by 202 respondents were subjected to quantitative analysis using descriptive statistics, factor analysis and Structural Equation Modelling (SEM).

1.7.4 Model Validation

The validation stage was conducted with the main objective of validating the technology transfer model developed in the primary study stage. The methods used in this stage were a focus group and case studies. The focus group membership consisted of participants from the relevant organisations responsible for the KVMRT technology transfer programme. The findings from the focus group also helped in fine-tuning the KVMRT technology transfer model. The case study approach is one of the primary methods for validating the developed technology transfer model. Five case studies were chosen as the platform for validating the technology transfer model. The data from these case studies were collected through interviews and the content analysis of related documents.

1.7.5 Findings and Recommendations

The final findings were prepared once the technology transfer model had been developed and validated for all the study stages. The recommendations made are for future similar projects that want to implement an effective technology transfer programme. The recommendations can be used by project owners to plan, develop and manage an effective technology transfer programme where the outcomes can be measured. The findings from the study can be used as a reference and guide for future research on technology transfer models and technology transfer processes used in large public projects, especially in rail infrastructure projects.

1.8 Thesis Structure

Below is a summary of each chapter and its purpose ranging from describing the research methodology, planning the study, the research implementation and the study findings. **Chapter 1** provides an introduction and background to the study. The chapter will provide the basis on the need for study and explain in detail its background. In the chapter, the objectives of the study are explained and how it will achieve the intended objectives. The chapter will also provide the problem statement of the study, its scope, the research stages and the thesis layout.

Chapter 2 provides a literature review for the study. The literature review covers previous studies related to a technology transfer process, technology management and technology transfer models. It covers the analysis of the literature covering technology transfer models and factors related to the technology transfer process. It also covers the types of technology transfer model and the evolution of government policy influencing the technology transfer model in Malaysia. The KVMRT conceptual model will also be developed in this chapter based on comparison analysis of selected technology transfer models found in previous studies and research.

Chapter 3 provides the details of the research methodology used to achieve the objectives of the study. The chapter will explain the data-gathering methods used for the study. It also outlines the research tools and analyses used for the conceptual model, primary study 1, primary study 2 and model validation stages.

Chapter 4 provides the results and analysis of the primary study 1 and primary study 2 stages. The results consist of the data gathered during the interview stage and the distribution of the questionnaire stage. It also explains the steps described and the justification for the analysis tools. Chapter 4 also provides details on the validation of the technology transfer model developed from primary study 1 and primary study 2. Gap analysis is also conducted in this chapter to meet the intended objectives of the study. The explanation in this chapter includes the validation methods and the result of the validation analysis.

Chapter 5 provides the findings and discussion of the results and analyses in Chapter 4. The chapter explains in detail the key findings and recommendations that can be adopted by similar future rail infrastructure projects to develop a successful technology transfer programme. The chapter also provides the findings from the gap analysis and the recommendations made to close these gaps.
Chapter 6 provides the conclusion and recommendations for the study. The conclusion includes a summary of the results and findings regarding the study's objectives. Recommendations for future research, as well as the limitations of the study are also explained in detail.

The questionnaire form used for the primary study 1 and the transcribed answers are attached in Appendix A and Appendix B.

REFERENCES

- A. Bakar, A. H. (2006). Capacity and Capability Development in Indigenous Construction Firms through Technology Transfer in Construction: A Malaysia Experience. Paper presented at the CIB W107 Construction in Developing Countries International Symposium.
- A. Khan, S. Z. (2011). Technology Transfer Effectiveness Through International Joint Ventures (IJVs) to Their Component Suppliers: A Study of the Automotive Industry of Pakistan. University of Birmingham, UK.
- Abbas, A. (2014). *The Characteristics of Successful Benchmarking Implementation* Massey University, Manawatu, New Zealand.
- Abd Rahman, N. S. a., Abd Razak, N., Ismail, K. and Ahmad Anuar, M. (2012). Innovation & Commercialization: Role of Technology Transfer Office in Malaysia Universities [Electronic Version],
- Abdul Mutalif, M. H. (2013). A Study on the Effectiveness of Technology Transfer Towards the Work Performance in SME's. Universiti Malaysia Pahang.
- Abdul Wahab, S., Che Rose, R. and Osman, S. I. W. (2011a). Measuring the Effects of Relationship Quality and Mutual Trust on Degree of Inter-Firm Technology Transfer in International Joint Venture. *International Business Research*, 4(3).
- Abdul Wahab, S., Che Rose, R. and Osman, S. I. W. (2012a). Defining the Concepts of Technology and Technology Transfer: A Literature Analysis. *International Business Research*, 5(1).
- Abdul Wahab, S., Che Rose, R., Uli, J. and Abdullah, H. (2011b). Relationships between Knowledge, Technology Recipient, Technology Supplier, Relationship Characteristics and Degree of Inter-Firm Technology Transfer. *European Journal of Social Sciences*.
- Abdul Wahab, S., Che Rose, R. and Wati Osman, S. I. (2012b). Exploring the Technology Transfer Mechanisms by the Multinational Corporations: A Literature Review. *Asian Social Science*, 8(3).
- Adams, A. and Cox, A. L. (2008). Questionnaires, in-depth interviews and focus groups. In P. Cairns (Ed.), *Research Methods for Human Computer Interaction* (pp. 17-34). Cambridge, UK: Cambridge University Press.

- Adzroe, E. K. (2015). A Study of E-Business Technology Transfer Via Foreign Direct Investment in the Ghanaian Construction Industry. The University of Salford, Salford, UK.
- Ahmad, R. and Sivanandam, H. (2017, 17 July 2017). Najib Launches Phase 2 of Sungai Buloh-Kajang MRT Line. *The Star*,
- Akubue, A. I. (2002). Technology Transfer: A Third World Perspective. *The Journal* of *Technology Studies*(28 (1)), 14 21.
- Alashwal, A. M., Abdul Rahman, H. and Radzi, J. (2014). Knowledge Utilization in Construction Projects: A Conceptual Framework. Paper presented at the Knowledge Management International Conference.
- Aldhahri, N. S. (2015). Investigating the ways of how e-learning enhances teaching and learning of computer science at tertiary institutions in New Zealand. Auckland University of Technology, Auckland.
- Ali Faizal, S. and Abdul Wahab, S. (2014). A Review on Technology Transfer in Context of Multinational Corporations. *International Journal of Business and Technopreneurship*, 4(3).
- Alizadeh, A., Chehrehpak, M. and Nazari-Shirkouhi, S. (2018). An empirical study on factors influencing technology transfer using structural equation modelling. *International Journal of Productivity and Quality Management*.
- Alshenqeeti, H. (2014). Interviewing as a Data Collection Method: A Critical Review. English Linguistics Research, 3(1).
- Anderson, J. C. and Gerbing, D. W. (1988). Structural Equation Modeling in Practice: A Review and Recommended Two-Step Approach. *Psychological Bulletin*, 103(3), 411 - 423.
- Andrenelli, A. and Gourdon, J. (2019). *International Technology Transfer Policies*. Pariso. Document Number)
- Aparecido Dias, A. and Silveira Porto, G. (2018). Technology transfer management in the context of a developing country: evidence from Brazilian universities. *Knowledge Management Research & Practice.*
- Aquino, A. and Succurro, M. (2011). Technology Transfer offices and academic spinoff creation: The case of Italy. *Journal of Technology Transfer*, 382 - 400.
- Arenas, J. J. and Gonzalez, D. (2018). Technology Transfer Models and Elements in the University-Industry Collaboration. *Administrative Sciences*, 8(19).

- Argote, L. (2011). Organizational Learning: From Experience to Knowledge. Organization Science.
- Asghari, M. and Rakhshanikia, M. A. (2013). Technology Transfer in Oil Industry, Significance and Challenges. *Proceedia Social and Behavioral Sciences*, 75, 264 - 271.
- Awang, A. H. A., Hussain, M. Y. H. and Abdul Malek, J. (2008). Promoting Knowledge Transfer in Science and Technology: A Case Study of Technology Park Malaysia (TPM). Zagreb: The Institute of Economicso. Document Number)
- Baden-Fuller, C. and Haefliger, S. (2013). Business Model and Technological Innovation. *Long Range Planning*, 46 424 -425.
- Baek, D.-H., Sul, W., Hong, K.-P. and Kim, H. (2007). A Technology Valuation Model to Support Technology Transfer Negotiations. *R&D Management*, 37(2).
- Bakar, M. H. (2015a). The Impact of Technology Transfer through Research and Devlopment Investment to the Financial Performance of SME's in Malaysia. Kobe University.
- Bakar, M. H. (2015b). Role of Technology Transfer through Research and Development to Increase Competitiveness. Do SME's in Malaysia Take Advantages on it? Kobe University.
- Balakrishnan, K. (2007). Evaluating the Effectiveness of Offsets as a Mechanism for Promoting Malaysian Defence Industrial and Technological Development. Cranfield University.
- Battistella, C., De Toni, A. F. and Pillon, R. (2015). Inter-organisational technology/knowledge transfer: a framework from critical literature review. *The Journal of Technology Transfer*.
- Bauer, S. M. and Flagg, J. L. (2010). Technology Transfer and Technology Intermediaries. *Assistive Technology Outcomes and Benefits*, 6(1).
- Beltrán-Morales, L. F., Almendarez-Hernández, M. A., Flores-Delgado, V., Trejo-Berumen, K. S., Lagunas-Vázques, M. and Ortega-Rubio, A. (2019).
 Technology Transfer Offices as Promoters of Technology, Innovation and Regional Development in Mexico. *International Journal of Innovation*.

- Bentahar, O. and Cameron, R. (2015). Design and Implementation of a Mixed Method Research Study in Project Management. *The Electronic Journal of Business Research Methods*, 13(1), 3-15.
- Birt, L., Scott, S. and Cavers, D. (2016). Member Checking: A Tool to Enhance Trustworthiness or Merely a Nod to Validation? *Qualitative Health Research*, 26(13), 1802 - 1811.
- Bradley, S. R., Hayter, C. S. and Link, A. N. (2013). Models and Methods of University Technology Transfer. Greensboro: University of North Carolina.
- Bui, N., Merschbrock, C. and Munkvold, B. E. (2016). A Review of Building Information Modelling for construction in developing countries Paper presented at the Creative Construction Conference.
- Burmistrov, A., Siniavina, M. and Illiashenko, O. (2017). Project Management Life Cycle Models to Improve Management in High-rise Construction [Electronic Version],
- Buys, A. J. and le Grange, L. l. (2012). A Review of Technology Transfer Mechanisms. University
- Cameron, E. H. (1960). Samuel Slater: Father of American Manufacturer. Portland, MA: The Bond Wheelright Company.
- Cascio, W. F. and Montealegre, R. (2016). How Tehnology Is Changing Work and Organizations [Electronic Version]. Annual Review of Organizational Psychology and Organizational Behavior,
- Cetin, D. (2009). Determinants of Technology Transfer in Developing Economies: The Case of Turkish Manufacturing Industries. Middle East Technical University.
- Chan, L. and Aldhaban, F. (2009). *Technology Transfer to China: With Case Studies in the High-Speed Rail Industry*. Paper presented at the PICMET 2009 Oregon, USA.
- Chege, S. M., Wang, D., Suntu, S. L. and Bishoge, O. K. (2018). Influence of technology transfer on performance and sustainability of standard gauge railway in developing countries. *Technology in Society*, 56, 79-92.
- Choi, H. J. (2009). Technology Transfer Issues and a New Technology Transfer Model. *The Journal of Technology Studies*, 35(No. 1).
- Coccia, M. (2019). What is Technology and Technology Change? A New Conception with Systemic-Purposeful Perspective for Technology Analysis. *Journal of Social and Administrative Sciences*, 6(3).

- Costello, A. B. and Osborne, J. W. (2005). Best Practices in Exploratory Factor Analysis: Four Recommendations for Getting the Most From Your Analysis. *Practical Assessment, Research and Evaluation, 10*(7).
- Da Silva, L. E., Filiz Karabag, S. and Berggren, C. (2017). Collaboration in international technology transfer: the role of knowledge boundaries and boundary objects. *Munich Personal RePEc Archive*.
- Dinakar, R. (2011). International Technology Transfers and the Role of Governments: A Study on Japanese Official Development Assistance to the Railway Sector in India. University of Tsukuba.
- Dixson, M. D. (2015). Measuring Student Engagement in the Online Course: The Online Student Engagement Scale (OSE). *Online Learning*, 19(4).
- Dominguez Gonzales, R. V. and Martins, M. F. (2017). Knowledge Management Process: a theoreticcal-conceptual research *Knowledge Management Process* 24(2), 248-265.
- Edge, T. (2013). MRT Corp to dish out RM4b jobs to local players. The Edge,
- Edler, J., Uyarra, E. and Yeow, J. (2013). Public Procurement as an innovation policy tool: choice, design and assessment. *Technological Forecasting & Social Change*, 86.
- Edmondson, A. and McManus, S. (2007). Methodological Fit in Management Field Research. *Academy of Management Review*, 32(4), 1155 - 1179.
- Elshahoupy, N. (2020). Review of the Strategies for Technology Transfer in Manufacturing Sector. *American International Journal of Business Management*, 3(3), 25-30.
- Estep, J. and Daim, T. (2016). A Framework for Technology Transfer Potential Assessment. Paper presented at the PICMET'16: Technology Management fro Social Innovation.
- Farahani, A. (2019). The Role and Influence of Government Economic Policies in the Process of Transferring or Developing the Required Technologies in the Aviation Knowledge-Based Industries and Aeronautics. *International Journal* for Modern Trends in Science and Technology, 5(4), 23-29.
- Farooq, R. (2017). A Framework for Identifying Research Gap in Social Sciences: Evidence from the Past. *The IUP Journal of Management Research*, 16(4).

- Fauzan, R. and Gooneratne, N. (2019). Working with the University Technology Transfer Office. Academic Entrepreneurship for Medical and Health Scientists, 1(3).
- Field, A. (2015). Discovering Statistics Using IBM SPSS Statistics 4th Edition: SAGE Publications Ltd.
- Galea, S. and Tracy, M. (2007). Participation Rates in Epidemiologic Studies. Ann *Epidimeol*, 17(9).
- Gibson, D. V. and Smilor, W. (1991). Key Variables in Technology Transfer: A Field
 Study Based on Empirical Analysis. *Journal of Engineering and Technology Management*, 8, 287 - 312.
- Gill, j. J. and Clark, M. (2010). Research Methods for Managers: SAGE Publications.
- Gliem, J. A. and Gliem, R. R. (2003). Calculating, Interpreting and Reporting Cronbach's Alpha Reliability Coefficient for Likert-Type Scales. Paper presented at the Midwest Research to Practice Conference in Adult, Continuing and Community Education.
- Groen, W. P. d., Lenaerts, K., Bosc, R. and Paquier, F. (2017). Impact of Digitalisation and the On-demand Economy on Labour Markets and the Consequences for Employment and Industrial Relations o. Document Number)
- Gunsel, A. (2015). Research on Effectiveness of Technology Transfer from a Knowledge Based Perspective. *Social and Behavioral Sciences*, 207, 777-785.
- Haghirian, P. (2003). Does Culture Really Matter? Cultural Influences on the Knowledge Transfer Process Within Multinational Corporations. Paper presented at the Eleventh European Conference on Information Systems, Naples. Italy.
- Haile, G. A., Srour, I. and Vivarelle, M. (2013). The Impact of Globalization and Technology Transfer on Manufacturing Employment and Skills in Ethiopia: Forschunginstitut zur Zukunft der Arbeito. Document Number)
- Hair, J., Tatham, R., Anderson, R. and Black, W. (1998). *Multivariate Data Analysis* (5th Edition ed.). London: Prentice-Hall.
- Hamdan, A. R. (2015). Assessment on the Effectiveness of Technology Transfer in the Klang Valley Mass Rapid Transit (KVMRT) Co-designing Offset Projects. Universiti Teknologi Malaysia.

- Han, S., Koo, D. D., Kim, Y., Kim, S. and Park, J. (2017). Gap Analysis Based Decision Support Methodology to Improve Level of Service of Water Services. *Sustainability*.
- Hassan, A., Jamaluddin, M. Y. and Queiri, A. (2015). *Technology Transfer Model For* the Information and Communication Industry. Paper presented at the Academics World 2nd International Conference, Kuala Lumpur, Malaysia.
- Hasselgren, B. (2013). *School of Architecture and Built Environment* KTH Royal Institute of Technology.
- Heslop, L. A., McGregor, E. and Griffith, M. (2001). Development of a Technology Readiness Assessment Measure: The Cloverleaf Model of Technology Transfer. *Journal of Technology Transfer*, 26, 369 - 384.
- Ho, S.-P., Tserng, H.-P. and Jan, S.-H. (2013). Enhancing Knowledge Sharing Management Using BIM Technology in Construction. *The Scientific World Journal*, 2013, 10.
- Hossain, M. M. and Kozubowski, T. (2014). A Novel Estimation Method Based on Maximum Likelihood. University of Nevada.
- Irwin and Moore. (1991). Technology Transfer and Communication: Lesson from Silicon Valley. *Journal of Information Sciences*, 17, 273 280.
- Isah, N. and Ali, M. (2015). A Review of the Literature on the Roles and Features of SMART Tunnel, Kuala Lumpur, Malaysia. *International Journal of Research* and Review.
- Ismail, K. and Ajagbe, M. A. (2013). The Roles of Government in the Commercialization of Technology Based Firms. *Middle-East Journal of Scientific Research*.
- Jafarieh, H. (2001). Technoloy Transfer to Developing Countries: A Quantitative Approach. University of Salford.
- Jagoda, K., Maheshwari, B. and Loseth, R. (2010). Key Issues in managing technology transfer projects: Experiences from a Canadian SME. Paper presented at the International Conference of Business Casess.
- Japan, E. o. (2016). The Malaysian Look East Policy. 2016, from http://www.my.embjapan.go.jp/English/JIS/education/LEP.htm
- Jomo, K. S. (1993). Industrializing Malaysia: Policy, Performance, Prospects. New York: Routledge.

- Jusoff, K. A., R, A. and Z. Takahashi, Y. (2009). The Japanese Influence in Malaysian Automotive Industry: Human Resources Management and Development Practices. *Management Science and Engineering*, 3(4).
- Kaur, S. (2016). SSP Line Project may cost up to RM40b, says MRT Corp. *The New Straits Times*,
- Khabiri, N. and Aslan Amat Senin, S. R. (2012). Identifying Main Influential Elements in Technology Transfer Proess: A Conceptual Model. *Procedia Social and Behavioral Sciences*, 40, 417-423.
- Khademi, T., Parnian, A., Garmsari, M., Ismail, K. and Lee, C. T. (2014). Role of Technology Transfer Office / Centre of Universities in Improving the Commercialization of Research Outputs: A Case Study in Malaysia. Paper presented at the Knowledge Management International Conference (KMICe) 2014.
- Khalifa, O. O. and Abdalla Hashim, A. H. (2012). Technology Transfer in developing countries. *Advance in Natural and Applied Sciences*, *6*(5), 620-624.
- Khayat, S. M. (2015). Factors Affecting Technology Transfer in the Phillippines Food Processing Industry. *Journal of Food Processing and Technology*, 6(5).
- Kiong, F. (2000). The Look East Policy: Its Impact in Promoting Japanese Management Techniques to Manufacturing Firms in Malaysia. Unpublished PhD Thesis, University of Stirling.
- Kline, R. B. (2011). *Principles and Practice of Structural Equation Modeling (Third Edition)*. New York: The Guilford Press.
- Kooli-Chaabane, H., Boly, V. and Yannou, B. (2014). Monitoring of Technology Transfer Projects in Industrial Clusters. *Journal of Innovation Economics & Management*, 73-91.
- Kumar, G. and Avery, J. L. (2018). *Effective Communication Portals for Technology Transfer* o. Document Number)
- Kundu, N., Bhar, C. and Pandurangan, V. (2015). Development of Framework for an Integrated Model for Technology Transfer. *Indian Journal of Science and Technology*, 8(35).
- L. Carroll, L. S. (2017). A Comprehensive Definition of Technology from an Ethological Perspective. *Social Sciences*.
- Lai, W.-H. and Chao, P. (2006). The Challenge of Technology Transfer in a Cross-Cultural Environment - From the U.S. to the Greater China Region. Paper

presented at the Second International Intelligent Logistics Systems Conference 2006.

- Lavoie, J. R., Kim, J. and Daim, T. U. (2017). A Technology Transfer Framework: A Case Study from the Energy Sector. *International Journal of Innovation*, *Management and Technology*, 8(4).
- Lee, L. and Sipalan, J. (2018). Malaysia, Singapore agree to defer high speed rail lin. *Reuters*,
- Lin, B.-W. and Berg, D. (2001). Effects of Cultural Difference on Technology Transfer Projects: An Empirical Study of Taiwanese Manufacturing Companies. *International Journal of Project Management*, 19, 287-293.
- Lin Say, T. (2012). Gamuda on track with MRT Project. The Star,
- Lowtan, D. M. and Sussman, J. (2004). *Rail System in Malaysia*: Massachusetts Institute of Technologyo. Document Number)
- Luggen, M. (2004). *Technology and Innovation Management in New Technology-Based Firms*. Swiss Federal Institute of Technology Zurich.
- M. Rogers, E., Yin, J. and Hoffman, J. (2000). Assessing the Effectiveness of Technology Transfer Offices at U.S. Research Universities. University of New Mexico.
- Maleki, A., Yazdi, N., Naeini, A. B. and Jassbi, A. (2017). Intermediate in Public Procurement of Innovation for Technology Transfer. Paper presented at the The 14th Asialics conference & the 7th International Conference on Management of Technology and Innovation.
- Malm, A. (2016). Technology Transfer within Related Offset Business From an Aircraft Production Perspective. Linkoping University, Sweden.
- Malmqvist, J., Hellberg, K., Mollas, G., Rose, Richard and Shevlin, M. (2019).
 Conducting the Pilot Study: A Neglected of the Research Process?
 Methodological Findings Supporting the Importance of Piloting in Qualitative
 Research Studies Intenational Journal of Qualitative Methods, 18, 1-11.
- Mandel, A. (2012). The Effects of Japanese Culture and History on the Adoption of Quality Control Techniques. Brandeis University.
- Mat Arifin, F. and Othman, N. (2018). The Dynamic of Policymaking Process in Malaysia. International Journal of West Asian Studies, 10(7).

- Mauricio, D. and Lopez, X. (2018). A Systematic Literature Review on Technology Transfer from University to Industry. *International Journal Business and Systems Research*, 12(2).
- Md Hasan, S. A., Othman, A., Mustafa, S., Wan Abd Wahab, W. N. Z. and Ismail, H.
 H. (2011). Development of Absorption Capability Attributes for Technology Transfer Performance: A Pilot Study in National Automotive Industry. Paper presented at the International Management Conference, Kuala Terengganu, Terengganu.
- Meulman, F. (2017). Challenges for Innovation Intermediaries in Technology Transfer : Forming Controls and Alliances. Eindhoven University of Technology.
- Mhd Sharif, S. and Ismail, Y. (2006). Technology Parks, Knowledge Transfer and Innovation: the Case of Malaysia's Information and Communication Technology (ICT) Small and Medium Enterprises. *International Journal of the Information Systems for Logistics and Management (IJISLM)*, 1(2), 133 - 142.
- MIGHT. (2009). Malaysia High Technology Industry Report 2009: MIGHT.
- MIGHT. (2014a). *Malaysian rail support industry Roadmap 2030*: Malaysian Industry-Government Group for High Technology
- MIGHT. (2014b). *MRT Offset Programme Success Stories* o. Document Number)
- MITI. (2009). Review of National Automotive Policy. Retrieved 11 May 2017, 2017, from http://www.miti.gov.my/index.php/pages/view/1449
- MOF. (2014). Policy and Guidelines on Industry Collaboration Programme in Government Procurement. Retrieved. from.
- Mohajan, H. K. (2019). The First Industrial Revolution: Creation of a New Global Human Era. *Journal of Social Sciences and Humanities*, 5(4).
- Mohamad, H. (2003). Rail Transportation in Kuala Lumpur. Japan Railway & Transport Review, 35, 21 27.
- Mohamad Hassan, N. A. H., Muhamad Noor, M. N. and Hussin, N. (2017). Knowledge Transfer Practice in Organization. *International Journal of Academic Research in Business and Social Sciences*, 7(8).
- Mohamed, M., Fung, H.-N. and Wong, C.-Y. (2015). Convergence Innovation in Railway Technology: How ERL of Malaysia attained its Co-evolution Structure for Systemic Development. *Asian Journal of Technology Innovation*, 92 - 108.

- Mohd Tobi, S. U. (2016). *Qualitative Research, Interview Analysis and Nvivo11 Exploration:* ARAS Publisher.
- Morton, S. M. B., Bandara, D. K., Robinson, E. M. and Carr, P. E. A. (2012). In the 21st Century, what is an acceptable response rate? *Australian and New Zealand Journal of Public Health*, 36(2), 106-108.
- MOSTI. (2016). Dasar Sains, Teknologi dan Inovasi Negara. Retrieved. from.
- MRTC. (2013). *MRT Corp Annual Progress Report 2013*. (M. Corporation o. Document Number)
- MRTC. (2015). MOU Signing Ceremony and Launching of KVMRT Training Centre. In M. Corporation (Ed.).
- Nachtigall, C., Kroehne, U., Funke, F. and Steyer, R. (2003). Why Should We Use SEM? Pros and Cons of Structural Equation Modeling. *Methods of Psychological Research Online*, 8(2), 1-22.
- Nee, E. A. (2013). Klang Valley MRT offset scheme to contribute RM3.5b: The Sun Daily.
- Nguyen, N. T. D. and Aoyama, A. (2015). The impact of cultural differences on technology transfer. *Journal of Manufacturing Technology Management*, 26(7), 926-954.
- Nightingale, P. (2014). What is Technology? Six Definitions and Two Pathologies. SSRN Electronic Journal.
- Oliveira, M. D. M. and Teixeira, A. (2010). *The Determinants of Technology Transfer Efficiency and The Role of Innovation Policies: A Survey*: Universidade Do Portoo. Document Number)
- Olsen, W. (2004). Developments in Sociology. Ormskirk: Causeway Press.
- Omar, R., Nawawi, A. H. and Takim, R. (2011). Measuring absorptive capacity in technology transfer (TT) projects. Universiti Tun Hussein Onn Malaysia.
- Omar, R., Nawawi, A. H., Takim, R. and Hassan, P. F. (2010). Technology Transfer (TT) and Technology Exchange (TE) in Malaysia. Paper presented at the International Conference on Education and Management Technology.
- Omar, R., Sallehuddin, A. M. and Tan, K. H. (2019). Technology Transfer in Mass Rapid Transport (MRT) Project in Malaysia. *Malaysian Construction Research Journal*, 8(3), 163-173.
- Omar, R., Takim, R. and Nawawi, A. H. (2008). Importing International Technology Through International Technology Transfer (ITT) Projects in Construction:

Synthesis of ITT Projects Models. Paper presented at the Transformation Through Construction.

- Omar, R., Takim, R. and Nawawi, A. H. (2012). Measuring of Technological Capabilities in Technology Transfer (TT) Projects. *Asian Social Science*, 8(15).
- Omar, R., Tan, K. H., Sallehuddin, A. M., Sarpin, N., Yahya, M. Y., Goh, K. C., et al.
 (2018). Implementation of Technology Transfer in Mass Rapid Transport (MRT) Project in Malaysia. Paper presented at the MATEC Web Conference.
- Ortiz-Gallardo, V. G., Tietze, F., R. Probert, D. and Phaal, R. (2014). Technology acquisition through collaboration: practical insights for technology suppliers. *International Journal Technology Intelligence and Planning*, *10*(1).
- Osano, H. M. and Koine, P. W. (2016). Role of foreign direct investment on technology transfer and economic growth in Kenya: a case of the energy sector. *Journal of Innovation and Entrepreneurship*.
- Osman-Rani and Piei. (1990). Malaysia's Industrialization and Trade: Issues, Options and Strategies. *Jurnal Ekonomi Malaysia*, 21 & 22, 13 44.
- Owusu-Manu, D.-G., Parn, E. A., Antwi-Afari, M. F. and Edwards, D. J. (2017). Modelling a Conceptual Framework of Technology Transfer Process in Construction Projects: An Empirical Approach [Electronic Version],
- Ozsungur, F. (2018). Adaptation Approach to Technology Transfer Strategy. *Afro Eurasian Studies*, 7(1).
- Palinkas, L. A., Horwitz, S. M., Green, C. A., Wisdom, J. P., Duan, N. and Hoagwood,
 K. (2016). Purposeful Sampling for Qualitative Data Collection and Analysis
 in Mixed Method Implementation Research. *HHS public Access*.
- Park, S. H. and Lee, Y. G. (2011). Perspectives on Technology Transfer Strategies of Korean Companies in Point of Resource and Capability Based View. *Journal* of Technology Management & Innovation, 6(1).
- Parke, H. S. and Sonesson, W. (2018). *The Key Aspects During Departmental Technology Transfer*. Uppsala Universitet.
- Parker, A. and Tritter, J. (2006). Focus group method and methodology: current practice and recent debate. *International Journal of Research & Method in Education*, 29(1), 23-37.
- PEMANDU. (2012). ETP Annual Report 2012. In PEMANDU (Ed.) (pp. 20 39).

- Phan, P. H. and Siegel, D. S. (2006). *The Effectiveness of University Technology Transfer, Foundation and Trends in Entrepreneurship* (Vol. Vol 2): Now Publishers Inc.
- Purushotham, H., Vaithianathan, S. and Shyam Sunder, C. (2015). Structural Equation Modeling (SEM) Approach to Identify Critical Success Factors of Technology Transfer: An Empirical Analysis from Indian Context. Advances in industrial Engineering and Management, Vol. 4(No. 2), 123 - 146.
- Ra, S., Shrestha, U., Khatiwada, S., Won Yoon, S. and Kwon, K. (2019). The rise of technology and impact on skills. *International Journal of Training Research*.
- Rajan, R., Dhir, S. and Sushil. (2020). Technology Management for innovation in organisations: an argumentation-based modified TISM approach. *Technology Management for Innovation*.
- Ramayah, T., Memon, M. A., Ting, H., Chuah, F. and Chear, J.-H. (2017). A Review of the Methodological Misconception and Guidelines Related to the Application of Structural Equation Modelling: A Malaysian Scenario. *Journal of Applied Structural Equation Modeling*.
- Rani, S. S., Rao, B. M., Ramarao, P. and Kumar, S. (2018). Technology Transfer -Models and Mechanism. *International Journal of Mechanical Engineering and Technology (IJMET)*, 9(6), 971 - 982.
- Rasiah, R. (1993). Free Trade Zones and Industrial Development in Malaysia. In K. S. Jomo (Ed.), *Industrializing Malaysia: Policy, Performance, Prospects*. New York: Routledge.
- Reddy, N. M. and Zhao, L. (1990). International Technology Transfer: A Review. *Research Policy*, 19, 285-307.
- Review, W. R. (2019). The Role of the Railway in Great Britain. Retrieved. from.
- Reynoso, C. F. (2010). The Use of Case Studies in Preparing Postgraduate Dissertations on Small and Medium Sized Firms. *Review of Business & Finance Case Studies, 1*(1).
- Rogers, E. M., Takegami, S. and Yin, J. (2001). Lessons Learned about Technology Transfer. *Technovation*, 21 (2001) 253-261.
- Rose, J. B. (1995). A Study of the Factors Associated with Successful Technology Transfer and Their Applicability to Air Force Technology Transfers. Air Force Institute of Technology.

- Ruiz, P. P. (2010). *Technology & Knowledge Transfer Under the Open Innovation Paradigm.* University of Bath.
- Sabbagh, K., El-Darwiche, B., Friedrich, R. and Singh, M. (2012). *Maximizing the Impact of Digitalization* o. Document Number)
- Safdari, M., Mehrizi, M. A. and Elahi, M. (2012). Effects of Imported Technology on Economic Growth in Iran. African Journal of Business Management, Vol. 6(24), 7120 - 7125.
- Sazali, A. W., Haslinda, A., Jegak, U. and Raduan, C. R. (2009). Evolution and Development of Technology Transfer Models and the Influence of Knowledge-Based View and Organizational Learning on Technology Transfer. *Research Journal of International Studies*(12).
- Sazali, A. W., Raduan, C. R. and Suzana Idayu, O. (2012). Defining the Concepts of Technology and Technology Transfer: A Literature Analysis. *International Business Research*, 5(1).
- Schmidt Bassi, N. S., da Silva, C. L. and Santoyo, A. H. (2015). Technology Transfer: An Interdisciplinary Process. *European Scientific Journal*.
- Schoonenboom, J. and Johnson, R. B. (2017). How to Construct a Mixed Methods Research Design. *Koln Z Soziol*, 69, 107-131.
- Secundo, G., Passiante, G. and Beer, C. d. (2016). Measuring University Technology Transfer Efficiency : A Maturity Level Approach. *Measuring Business Excellence*, 20(3), 42-54.
- Segman, R. (1989). Communicating technology: An historical view. The Journal of Technology Transfer, 14(3), 46-53.
- Shadfar, S. and Malekmohammadi, I. (2013). Application of Structural Equation Modeling (SEM) in Restructuring State Intervention Strategies Toward Paddy Production Development. *International Journal of Academic Research in Business and Social Sciences*, 3(12).
- Smith, B. T. (1995). *The Process of Technology Transfer: A Case Study of the National Aero-Space Plane Programme*. Air Force Institute of Technology.
- Smithson, J. (2000). Using and analysing focus groups: limitations and possibilities. *International Journal Social Research Methodology*, *3*(2), 103-119.
- Sonmezturk Bolatan, G. I., Gozlu, S., Alpkan, L. and Zaim, S. (2016). The impact of technology transfer performance on total quality management and quality

performance. Paper presented at the 12th International Strategic Management Conference, Antalya, Turkey.

- SPAD. (2014). National Land Public Transport Master Plan. Retrieved. from.
- Spann, M. S., Adams, M. and Souder, W. E. (1995). Measures of Technology Transfer Effectiveness: Key Dimensions and Difference in Their Use by Sponsors, Developers and Adopters. *IEEE Transactions on Engineering Management*, 42(1).
- Starman, A. B. (2013). The case study as a type of qualitative research *Journal of Contemporary Educational Studies*.
- Stoyanov, T. (2017). *Knowledge Transfer Digital Transformation*. Paper presented at the EDULEARN17 Conference.
- Taherdoost, H. (2016a). Sampling Methods in Research Methodology; How to Choose a Sampling Technique for Research. International Journal of Academic Research in Management, 5(2), 18-27.
- Taherdoost, H. (2016b). Validity and Reliability of the Research Instrument; How to Test the Validation of a Questionnaire/Survey in a Rsearch. Intenaional Journal of Academic Research in Management, 5(3), 28-36.
- Tahri, A. (2012). *Islamic Civilization's Contribution to Science and Technology*. Paper presented at the 2nd Asia-Arab Sustainable Energy Forum.
- Tavakol, M. and Dennick, R. (2011). Making Sense of Cronbach's Alpha. International Journal of Medical Education, 2, 53-55.
- TDA. (2016). ICP History and Background. Retrieved 1/10/2018, 2016, from http://tda.my/icp/icp-history-background/
- Teegavarapu, S. and Summers, J. D. (2008). *Case Study Method for Design Research: A Justification.* Paper presented at the IDETC/DTM 2008, New York, USA.
- Teijlingen, E. R. and Hundley, V. (2002). The Importance of Pilot Studies. *Social Research UPDATE*(35).
- Then, K. L., Rankin, J. and Ali, E. (2014). Focus Group Research: What is it and How Can it be Used? *Canadian Journal of Cardiovacular Nursing*.
- Thi Canh, N., Thanh Liem, N. and Vinh Khuong, N. (2019). Th Impact of Innovation on the Firm Performance and Corporate Social Responsibility of Vietnamese Manufacturing Firms. *Sustainability*.

- Thohari, H., Ayu, P., Indria, T. and Sueb. (2013). *The Development of Technology for Human Civilization*. Paper presented at the The Third Basic Science International Conference.
- Thompson, N., Fearnhead, N. and Claase, R. (2017). Digitalisation in Construction -Industrial Strategy Review, supporting the Government's Industrial Strategy.
 (A. Limited o. Document Number)
- Toner, P. (2011). Workforce Skills and Innovation: An Overview of Major Themes in the Literature. Paris: Centre for EDucational Research and Innovation (CERI)o. Document Number)
- Tunca, F. and Nur Kanat, O. (2019). Harmonization and Simplification Roles of Technology Transfer Offices for Effetive University - Industry Collaboration Models. *Procedia Computer Science*, 158, 361-365.
- Udochukwu Frank, I., Rong, D., Batool, H., Ikechukwu Paul, I. and Emmanuel Tungom, C. (2017). Technology Transfer in Construction and Management: A Case for Nigeria Construction and Management Sectors. *Management Science* and Engineering, 11(2).
- UNCTAD. (2014). Studies in Technology Transfer: Selected Cases from Argentina, China, South Africa and Taiwan Province of China. (U. N. Publication o. Document Number)
- Ungureanu, M., Pop, N. and Ungureanu, N. (2016). Innovation and Technology Transfer for Business Development. Paper presented at the International Conference on Manufacturing Engineering and Materials, Slovakia.
- Uusitalo, P. and Lavikka, R. (2020). Technology transfer in the Construction Industry. *The Journal of Technology Transfer.*
- Uyarra, E. and Flanagan, K. (2009). Understanding the Innovation Impacts of Public Procurement. *European Planning Studies*.
- Van Norman, G. A. (2017). Technology Transfer: From the Research Bench to Commercialization. *JACC: Basic to Translational Science*, 2(2).
- Vega-Gonzalez, L. R. and Roberto Misael, V.-S. (2018). Overview Case Analysis Applied to Evaluate Technology Transfer Projects of a Mexican Public University. *Journal of Technology Management & Innovation*, 13(3).
- Vitanov, N. K. and Ausloos, M. R. (2012). *Models of Science Dynamics Encounters Between Complexity Theory and Information Sciences*.

- Wad, P. and Govindaraju, C. V. G. R. (2011). Automotive Industry in Malaysia: An Assessment on its Development. Int. J. Automotive Technology and Management, 11(2).
- Waroonkun, T. (2007). Modelling International Technology Transfer in Thai Construction Projects. Griffith University.
- Weston, R. and A. Gore Jr., P. (2006). A Brief Guide to Structural Equation Modeling. *The Counseling Psychologist*, 34(5), 719-751.
- Williams, F. and Gibson, D. V. (1990). Technology Transfer: A Communication Perspective. Newbury Park: Sage Publications, Inc.
- Wiseall, S. S., Kelly, J. C. and Kelly, T. P. (2001). *Transferring Design Research into Rolls-Royce*. Paper presented at the International Conference on Engineering Design.
- Wolfgang Thurner, T. and Zaichenko, S. (2015). Sectoral Differences in Technology Transfer. *International Journal of Innovation Management*, 20(2).
- Wong, A., Ng, P. and Lo, M. F. (2018). Factors that affect the acceptance of new technologies in the workplace: a cross case analysis between two universities. *International Journal of Education and Development using Information and Communication Technology*, 14(3), 209-222.
- Xu, M., David, J. M. and Kim, S. H. (2018). The Fourth Industrial Revolution: Opportunities and Challenges. *International Journal of Financial Research*, 9(2).
- Yauch, C. and Steudel, H. (2003). Complementary Use of Qualitative and Quantitative Cultural Assessment Methods. *Organizational Research Methods*, 6(4), 465-481.
- Yeasmin, S. and Rahman, K. F. (2012). "Triangulation" Research Method as the Tool of Social Science Research. *BUP Journal*, *1*(1).
- Yong, A. G. and Pearce, S. (2013). A Beginner's Guide to Factor Analysis: Focusing on Exploratory Factor Analysis. *Tutorials in Quantitative Methods for Psychology*, 9(2), 79 - 94.
- Young, T. J. (2015). Questionnaires and Surveys
- [Electronic Version],
- Yusoff, I., Ng, B.-K. and Azizan, S. A. (2021). Towards Sustainable Transport Policy: A Rail-based Transit System in Klang Valley, Malaysia *Plos One*.

Zaidi, M. A. and Davies, H. (2011). Knowledge Transfer: A Model Framework for Construction Knowledge Integration. Paper presented at the 36th Australasian University Building Educators Association (AUBEA) Conference.

Appendix A Question Form (Primary Study 1)

TECHNOLOGY TRANSFER MODEL FOR THE KVMRT LINE 1: SG BULOH TO KAJANG

INDUCTION INTERVIEW QUESTIONS

NAME OF INTERVIEWEE : POSITION :

YEARS EXPERIENCE IN TT PROCESS :

NO.	QUESTIONS	ANSWERS		
1.	CONCEPTIONS OF TECHNOLOGY			
А.	What are the agency's requirements or			
	objectives for Technology Transfer?			
В.	What are the strategy or initiatives used by			
	the agency to identify the requirements for			
	Technology Transfer?			
C.	What are the current Technology Transfer			
	Programme or mechanism used?			
2.	TECHNOLOGICAL ACTIVITY AND TT TRANSF	ER		

Α.	Please explain the agency's process in	
	terms of planning for Technology Transfer	
	Program/Projects?	
В.	Do you put any proper mechanism to	
	monitor the activities?	
2	COMMUNICATION CHANNELS	
J.	What are the current communication	
А.		
	channels for Technology Transfer process	
	and activities?	
В.	Does the communication channels	
	effective and understood by all the related	
	stakeholders?	
C.	If not, what are the improvement you can	
	suggest to improve the communication	
	channels?	
4	MODES OF TECHNOLOGY TRANSFER	
 	What are the available Modes of	
д.	Tochnology Transfer in the agong 2	
	Technology Transfer in the agency?	

В.	What are the most used modes of	
	Technology Transfer?	
6		
C.	which modes in your opinion is the most	
	effective?	
5.	IMPACTS/RESULTS OF TECHNOLOGY TRANS	SFER
А.	What do you consider as the criteria for the	
	results of Technology Transfer in the	
	project?	
В.	What are the measurement used by the	
	agency to measure Technology Transfer	
	results?	
C.	What are the innovation resulted from the	
	Technology Transfer process?	
6.	GAPS IN THE TECHNOLOGY TRANSFER PROC	CESS

Α.	What a	re the Le	ssons Learn	ed from	the	
	previous	evious technology transfer process?				
В.	What a	re furthei	r improveme	ent that	you	
	can suga	gest to teo	hnology trai	nsfer pro	cess	
	that have	o hoon in	nlomontod)		
	that hav	e been in	ipiementeu			
7	FACTOR					R
7.						n
А.	Does th	is factor a	ffects?			
	1	2	3	4	5	
	Very	Low	Moderate	High	Very	_
	Low				High	
		-			1	
	Transfei	r Environr	nent			
	- (Complexit	v of Technol	0gv =		
		lompicati	y of reennor	ову – <u> </u>		
	- (Constructi	on Modee.g	.eg. JV vs	Sub-co	ontracting) =
		.				
	- (Jrganizati	on Policy =			
	- (Governme	ent Policy =			
	- Government Enforcement =					
	Learning	Fnviron	ment			
	Leannig		·			
	 Relationship between WPCs and Recipients = 					
	- 1	Mutual Tr	ust =			
		-		-		

-	Clear Understanding of the scope of technology transfer =					
-	- Effective communication =					
-	Strong commitment from senior management of WPCs and Recipients					
	=					
-	Teamwork =					
-	Supervision =					
WPCs	WPCs Characteristics					
-	Willingness to implement =					
-	Degree of Experience =					
-	Appropriate Management practices and approaches =					
Recip	Recipient's Characteristics					
-	Willingness to learn =					
-	Experience working with Foreigners =					
-	Appropriate Management practices and approaches =					
-	Adequate knowledge base =					
Outco	omes of TT					
Econo	Economic Advancement					
-	Firms competitiveness in the market =					
-	Firms overall performance (e.g. better management, better quality)					
	=					
-	Firms overall profitability =					
Know	Knowledge Advancement					
-	Improve workers knowledge and skill =					
-	Enhance their working practices =					
-	Improve firm's skill base over the long term =					

	 Increase the innovation in the firm = 				
	Project Performance				
	 Enhanced the project financial performance (e.g. project budge cashflow) = 				
	 Enhanced schedule performance (e.g. on time completion, resolution) 				
	 Enhanced the project quality standards (e.g. less re-work, improved deliverables) = 				
В.	Can you propose any other Factors affecting				
	Technology Transfer for the organization?				

Appendix B Transcribed Interviews (Primary Study 1)

EXPERT INTERVIEWEE CONSENT FORM (TEMPLATE)

:

Title of Project

Model of Technology Transfer in the Klang Valley Mass Rapid Transit Project

Name of Researcher :

Abdul Rahman Hamdan

Contact of Researcher:

Email :

H/P :

Statements	Please tick where		
	appropriate		
	Yes	No	
I have read the transcribed sheet from my interview session			
for the above research and acknowledge it is true			
I agree to take part in the research process as a respondent			
for the study			
I understand that my participation in this research is			
voluntary			

Name of Expert Interviewee:

[Name of Expert]

[Position in the Company]

I hereby agree to take part in this research

Signature :_____

Date :_____

Research Supervisor

Dr Mohamad Syazli Bin Fathi

Universiti Teknologi Malaysia

Razak School of Engineering and Business Management

Respondent A1

Have to see in which angle you see. We have system, civil and design and build. Which angle are you looking. If you consider this as construction phase, the technology transfer more towards system and underground. Technology transfer for underground happens even during preliminary design stage. For the Tunnel Boring Machine design, the technology transfer is during the concept design itself. I would say most technology transfer happens during the manufacturing phase. During design stage we do sent people to Thailand for Signalling system and China for Electric Train to learn but most of the Technology Transfer happens during manufacturing. Detail design I don't think so because most of the design is proprietary. It is more to the commitment by the WPC. The TT is dependent on WPC and Recipient.

In terms of objective, MRT follows the Offset Requirement Document (ORD) set by the government for the technology transfer. But, MRT is committed to comply regardless not have to follow the ORD. We don't get fund from government. No internal objectives. MRT use the ORD as the guideline. MRT have offset programme, Bumiputera mentoring programme.

Planning for TT depends on the schedule A set in the ICP programme. We appoint dedicated consultant to monitor. We have the ICP Committee Meeting every quarterly as well as the working committee to monitor the TT programme.

In terms of communication channel, procurement is playing the lead and we will coordinate with the technical, HR or etc. Centre point and reporting by procurement. System team will brainstorm on what project they want to do and procurement will facilitate. You have training, OJT so each unit do their own thing, but procurement consolidate. HR is more to resource planning and appointment of new staff.

The communication mechanism is effective. Will improve better with no intervention from external parties. We should have the benefit of having our own staff to benefit but due to external parties, other people are benefiting so the benefit is loss.

Classroom, OJT, overseas training. Classroom mode is the most used. The most effective is On-the-job training. OJT is the best but you cannot accommodate all. There is limitation to participants.

Criteria for a result of TT is job enhancement and knowledge enhancement. Some of the trainee went to join the operator. So, it is more to career progression and improvement. Other than that no. Another criteria can be publication from universities based on data shared to them.

The measurement for TT is that they are marketable and employability. Improvement to their salary. They should be better planning for the programme. Initial planning is a bit of shame. In other things, MRT always plan from the beginning. Unfortunately for the Technology Transfer programme, there is no planning at the beginning as no one knows the best way to do this. We know the gap industry have and what we can commit to serve. I wish we have thought this programme properly so that it can meet industry's need. The planning should have a common ground so that the recipient involved can contribute back. MRT, industry players and operator should have contributed to the IRD at the very beginning. That one is lacking. If you look at the IRD and ISD document, it is still very general and you do not know what to do still. By right, it should be more specific, for example, for signalling, how many signalling engineers

you want to develop as we are lacking signalling engineers. We still depend on the expatriates. In terms of technology, manpower and resources they should have identified during the planning.

The planning we do currently is based on the proposed schedule A in which is from WPC. So, we are dependent on what WPC are telling us instead of us telling them what to do. The consultant should be more intelligent proactive and give more specific. The IRD should have been a consolidated effort between the government agencies, regulatory body, industries and the operators. This is still lacking. Think tank is not there. Factor that can be included that is not here is third party interference.

Respondent A2

All the trainees for the offset programme in Line 1 has been absorbed for Line, 2. All have been employed by MRT. For MRT requirement, technology transfer related to technical team, so we request Head of Department of respective Systems team what are the requirements such as at what level, what subject and scope. We have that session requesting them the requirements. The offset requirement document given by the government is only for the tender stage. Once a contractor selected, the WPC's technology transfer proposal then is the aligned with MRT's policy and direction. Example, does the company allow to take new fresh graduate for the technology transfer programme or not. And then what is the career path for this graduates, where are putting them after this. Or maybe, we just want to enhance the existing staff capability that already here at MRT.

This direction can be determined by the project team and also aligned to HR. HR also has their own strategy and career path plan for the staff. Regardless if there is an offset or not, our internal HR already have technology transfer plan based on the expatriates that we hired. They have to share their knowledge to local MRT staff. One way formal channel is through presentation. We invite any staff interested to attend the presentation by this expatriates. So, HR encourage the expatriates to transfer their knowledge to MRT local staff.

Other ways of technology transfer is the local staff is working under this expatriates. These local staff will be exposed to the way how this expatriates work and that is considered as technology transfer itself. It is like a mentor-mentee programme. It is not a structured programme.

In terms of OJT and presentation channel, most effective is the OJT. Presentation is too wide and general. They only explain the history, what type of technology available and not that specific. But with OJT, you can have troubleshooting problems and they can learn that on site. OJT is better.

The main technology transfer channel for MRT is the offset programme and the expatriate's knowledge transfer programme. In terms of planning for technology transfer, I think the rail system technology is very niche and specialised knowledge. If HR try to recruit locally but no available local talent, HR will try to find foreign. So, the strategy of HR is mainly to fill the gap for delivering the project.

In my opinion, if MRT really serious in replacing the expatriates someday with the local, the expatriates should act as the advisor only and not put them as director. Example, put local people as tunnel director and the expatriate act as advisor to the director. This expatriate will train this local to be competent as him. This way they will actually be taking the responsibility and doing the job. But the problem is that MRT project is a fast track and we don't want to take risk for the project. The limitation is time. This thing cannot be a trial and error. If we take local and try to train but we cannot compromise the project.

If we talking proper monitoring is the consultant MRT appointed, the ICP Management Unit. The MRT project team needs to monitor the project. The monitoring committee is the MIEC, IWC. Because of offset, we have meeting and progress update. There is a structured approach. Communication is through progress meeting and committee. It should be effective. Cannot think any other effective platform besides that project management.

To improve, must have working group involving the HR, project team and procurement itself. Act as interfacing meeting. PDP also can get involved. It can be better.

Difference between Line 1 and Line 2 is that only selected packages are selected for technology transfer. So, we missed certain packages that should have technology transfer. For Line, 2, it is different as we based on certain value.

Mode of technology transfer is classroom, OJT, mentor-mentee, knowledge sharing via presentation with small group and large group. For example I went to over sea, when I get back I have a knowledge sharing session with my colleague on what I learned.

The most used technology transfer channel is OJT. The most effective is OJT. Due to hands on. Trouble shoot and problem solving real problems.

Result of the technology transfer is developing engineer in terms of competency and exposure from the real project. For instance, there is no signalling engineer before. Now, the engineer is specialised to a new field. There is also salary increase and promotion. Before trainee and now become engineers.

There might be difference compared to graduates who did not go for the technology transfer programme and graduates who participate in the programme. Because the graduates is now highly specialised. The graduates who participated in the technology transfer is reliable as they have undergo a steep learning curve.

The measurement used to measure the competency of the engineers is through yearly performance appraisal and also half year review.

In terms of innovation, there is not much can be seen. Offset is something new. Maybe not in Line 1. But maybe long term we can only see innovation. In order for innovation to happen, they must have personnel that is very experienced and good knowledge. Human resource is important and normally small companies cannot afford to hire expertise for long term. If you want to do R&D, the cost is high but the return is very minimal in the short term. So, if you are a small company you do not have the funding to do R&D based on the offset programme. Only big companies can afford to do R&D because they have big pockets.

The offset trainee is much more versatile. Trainee from ET not necessary work in ET package. They can work under COMMS package and they able to adapt.

Lessons learned from Line 1, planning and actual thing might be different. For example, MRT supposed to be the operator but that did not happen. Now we know we could not take many staff. Another, if IMU wants to be as the champion for the national roadmap, they should make known to other stakeholders like MRT and Prasarana on what are their objectives and target. The target is not clear. They want to develop human capital or local industry? That is not clear. It is either they have a detail roadmap but not make known to us or the current roadmap is too general. They must make it more specific. When too broad, you will try to do everything so you cannot focus. For example, MRT you need to focus on this, Prasarana you need to focus on this, so that we can achieve more and not duplicating.

Another improvement is that Malaysia needs to see the technology transfer success stories such as foreign signalling companies in Bangkok. Even without offset they can collaborate with universities. Maybe there is incentive for them to do technology transfer programme. So, this company are willing to transfer technology. It is a win-win for them. They do not need bring in expatriates as they can employ local and they can get business from that country. It should be on a win-win situation. Attract foreign investment and train local. Now, in Malaysia it is only project based. Once, project finished than they left. Foreign signalling companies can set a base in Bangkok for long term. The government agency is the one should be leading this and MRT will follow.

Another factor needs to be included is the rail academy. Established higher learning specific for rail support industry. It is the learning infrastructure for technology MRT. This includes having trainers and lecturers. This can accelerate the technology transfer, more structured and can be more innovative.

The effectiveness is when the young engineers is attached to the real environment, working alongside the engineers who is doing the actual work. They got not only theory but also handson practical. Not all knowledge can be transferred in classroom. Some of the knowledge they can pick up during OJT. More effective if during attachment, the WPC actually give real jobs to the trainees. Trainees to assist WPC indirectly the trainees will benefit more. I remember one trainee attached at Bombardier office doing interface CSD and SEM drawings. Even though they are using the trainees to do the review, in the same time the trainee can learn.

A good attachment if we give the attachment 100 percent to the WPC. The WPC take care of the trainees. They treat the trainees as a staff so they will train the trainees to do their work. They will make sure the trainees do their work properly so the trainees will learn better. If the trainees are attached at MRT it will not be that effective.

During Putra LRT, it is different concept. When I joined as an engineer, the direct OEM provide the training. For example the auxiliary power unit, the supplier is a British OEM, they provide the training. It is classroom plus OJT.

The planning of technology transfer is based on the Offset Requirement Document, in which MRT did provide some input to it. The input is from MRT. How many engineers and how long. If there is no ORD, we follow the contract. Under the contract, WPC needs to conduct train the trainer programme and training. The project team set the training requirements in the contract.

In contract deliverables, the training is for the operator and maintenance can maintain the system once WPC handover. OEM do the training just for maintenance and operation. When come to Offset, we should ask more from WPC. Under contract deliverables, they provide level 1 and level, 2 training. But under Offset, we can request up to level 3 maintenance training. So, far we have never capture knowledge during design stage. Only the design is ready, the training is provided. However with Offset, we can have knowledge transfer during design.

Mechanism used for TT is Offset programme and contract deliverables. We plan based on my past experience in Kelana Jaya Line, we know how many engineers required to perform maintenance.

In Offset, we monitor through the consultant appointed. For contract deliverable, we have proper record and evaluation form. The communication channel is ok because we have the MRT IC Committee and the working committee.

Line 1, we do interview. We appointed panel for evaluation including HR and project team. Project team decides the requirement and HR will plan the resources.

One improvement I can suggest is how to measure the competency of the participant. Need proper mechanism. Supposed to be led by HR or by the consultant, IMU. Based on Line 1, the competency is proposed by WPC so there is no standard guideline on the competency level need to be reached. How to measure the effectiveness of the project. Does the participant really receive the knowledge? IMU currently only monitor the progress.

Collectively, the level of competency should be agreed by all. Mode of TT - OJT, Classroom. Offset engineers is involved in testing and commissioning. The real implementation. They are now independent and no longer OJT. For signalling, we assign them to do reports/assignment. Mostly used – OJT and Classroom. Most effective – OJT. Criteria result – Competency.

Currently there is no proper competency measurement. In real world, they can now work independently no need supervision anymore. How do measure the result - Observation and Yearly performance job appraisal. However, this measurement should only be done by project team.

No innovation but only competent engineers. They can review design documents.

Lesson learned we must thoroughly review proposal by WPC. Training module by WPC. Otherwise we only received basic technology transfer. In Line 1, due to short time. WE leave it to WPC. For Line, 2, we propose WPC to propose earlier. We want more advanced technology transfer. Also another lessons learned, we do not need many participants so that the programme can be more effective. We need to consider WPC limitation such as space, trainer availability, facilities. For signalling is only 4 staff attachment and their scope of TT is clear. We attached them to a specific area. WPC can train them one-on-one.

I suggest for the TT, we should give to experience engineer as well. For fresh graduates, WPC only give basic TT. But if we want enhanced technology, we must give them experience engineer more than 5 years.

The place for technology transfer is very important. WPC needs to provide the facilities for effective TT.

Respondent A4

During design, our staff is also attached to WPC. Technology transfer happens mostly in construction phase that includes design. Technology transfer has many categories of transfer. Most transfer is effective. For example, in the case of MRT project, the WPC appointed a local company as offset recipient. So, these people in the local company is already involved in preliminary design. The highest technology transfer happens during detail design. Construction is where they implement it.

There are sometimes the provider are keen to transfer but the recipient are not ready. For example like this local company where they are not serious and take it lightly. Other reason is the local recipient do not have sufficient people. Some of the local recipient use the programme as just to get them jobs instead of using it to develop them.

The technology transfer can happen in a meeting, sometimes in a classroom or sometimes in a form of On Job Training (OJT). The commitment to TT must be equally between Provider and recipient. For example, like one of the local recipient, they are eager for the transfer, it has been a win-win situation for both parties. As for the technology transfer environment, if the recipient are eager for technology transfer, they themselves would create that environment, for example, they organize workshop, they go to Japan, send their people, that shows that they are committed. Because technology transfer environment is subjective, but if both are committed, the right environment will happen.

There government policy has no effect to the learning environment. It is up to the WPC and recipient to set the learning environment. Government policy is not a factor influencing the learning environment. The government policy is there only to make sure the WPC under their offset obligation. The key factor is still on the commitment from WPC and recipient. The government policy is more to a guideline and it is not a critical factor. They only need to fulfil the offset credit value. But successful is still subjective. It depends on the interpretation of both WPC and recipients.

Our technology transfer objective is more to know-how and technical knowledge. We are consider new company we need people to manage the project well. In the railway market, is always a shortage and we have no choice to train our own people. So, we identify which is the area as a gap and we put people to train there. For Line 1, the objective is more to human capital development. We employ more than 30 to 40 of young engineers to go for the offset project. The objective of the technology transfer is actually defined together with MIGHT at that time. MIGHT already have the Gap analysis and rail blueprint. As a complement to that to look at area MRT requires for technology transfer especially knowledge transfer. If you look what relates to MRT is more to knowledge transfer. But if you look at the KVMRT TT programme, it is more to the industry requirement. During tender stage, I did asked whether MIGHT perform any gap analysis. So, we use the blueprint as a guide.

The mechanism for TT are Offset and OJT. People who are not in the offset programme use OJT as a mean to get the knowledge transfer from contractor.

OJT is planned during the construction period. We plan the placement people according to the progress of the project from preliminary design, detail design, construction, installation and testing. Even without offset we will do that.

This staff needs to give weekly report and from there we gauge on how much knowledge they have acquire in that particular session. If they attach during detail design, the technology provider will provide the assessment. It is done weekly and based on that, WPC as the technology provider will submit a report to MRT.

We engage HR as well, we call all the staff who is involved in the offset programme and they present what they learn. Three department is involved, procurement, project team and HR. HR is interested in the development of the people and how we run the training and the knowledge acquired. Both HR and project team set the requirement. HR have its own KPI for their training. Area for development is set by project team. We say this is the gap and this is the area this engineers to be train to. In terms of successfulness, the project team will decide whether the knowledge transfer is successful or not. Project team understand more on the technical part of this. All of the parties understand their role. Procurement is more to the management aspect of it. Offset project is handled by the MRT procurement. But the actual beneficiaries are the project team.

Improvement to the communication.

Improvement in terms of communication, when we started, not many people knew the objectives of the programme. Towards the end of the programme people now understand. To gauge its success or failure is very objective. When the people go on Line 2 we can know whether they are successful or not. These engineers involved from design to construction and they come from zero experience in rail. Now they can even give feedback and input back to their knowledge provider. Advantage to people who involved in Offset is the experience gained.

Types of TT are Classroom, OJT. They get involved in the design process, e.g. Comment on the document. They get to be involved in the process even though at that time they do not know the right answer. They have to read, discuss and through meetings they know the answer and the technical terms.

They did a simulation study at Japan that includes FAT also. Other channel is self-learning. Example, here they learn 750 DC for MRT. But to be good engineer, they need to know other systems. So, they are asked to go visit ERL, KTM, Ampang Line. So what are the difference and what are the similarities. They need to know the reasoning on why KTM using this system, why ERL using this system.

The most effective channel is OJT. For me the result of TT is for the engineer to know the system and for them to challenge it. They are able to have critical thinking and they can argue. They can argue with the designer on why it is done like this. In terms of MRT, better manage their project, better quality system. This engineers are exposed to overseas practices and they can propose the best system.

Currently there is no exact measurement for TT. In terms of during assessment, we have questions and answer. Similar like exam. The actual how much they absorb we cannot quantify at the moment. But it is useful for them to be able to comment for the next project. Do system improvement. Performance appraisal can also be used as measurement tool also.

Some of the lessons learned is that the recipient must be ready. Don't use this opportunity of TT as a way to get a job. They supposed to take it positively. Probably it is against the norm

what they normally do for a project. They need to improve. If they do not have sufficient manpower, they need to get the manpower for the TT. That is in terms of product technology transfer but in terms of training, actually the recipient itself plays the important role. If the trainee is lazy and do not want to learn, do not want to be involved, how much they can grasp. This is very much on the trainee, how much they can grasp the knowledge. Attitude of the recipient is important. When we hire this people, we choose them carefully. They are the chosen one. If they have attitude problem, the programme will fail.

In terms of improvement to TT programme, from management point of view, we need to clearly identify the gap and the industry requirement. MRT and also the government body such as MIGHT and TDA. There a lot of rail project, it is a waste if we do not capitalize. Training and human capital development, human capital development now more focus on enhancing junior engineer. How about the senior engineer? Middle and upper engineer. They still hire foreigner for this position. We need a programme for the middle one. For Line, 2, we do not want to focus on junior engineer. We are targeting development for intermediate and advanced course. Line 1 is more entry / basic course, they now have experience and we targeting for advanced course. Now, there is no clear programme how we can replace the expatriate. This should be at MRT and TDA level. Not just MRT but around the world have a shortage in rail expertise. rail support industry is a government led industry. The more complex the less effective the technology transfer.

Joint-Venture is much more effective as they will do it together and that is more effective. Is enforcement an important factor? It can be an effective tool to force WPC to transfer technology. Senior management commitment to the programme is very important.

Look at category of TT. If HCD, the knowledge that they received is already a reward. Training itself is a reward to the recipient. You need to put category to the TT. Category in terms of human capital development or training. Another category is product knowhow. If you want to assist and develop new product. The assessment is different. You cannot use the same measurement for different type of TT. If you can separate that you can have more accurate data.

If HCD, OJT is more effective. But product development, JV is more effective. In power offset, both types of TT is there. The local recipient in the KVMRT power supply package is ok because they have experience, good relationship and trust with the technology provider. All the factors plays an important role. For some of the local recipient, they burned the trust and relationship with their technology provider and that is why it is a problem. Criteria mentioned here does happen in actual MRT TT programme.
Respondent A5

There concept design first before preliminary design. TT happens mostly during construction. If we look at ET package, during detailed design they decide on the suppliers. If we look at construction, the TT is not that significant because the suppliers use their own technology. If the design by WPC and advised by WPC is ok, but for the project is only business arrangement. In the local assembly plant, there is TT in the process of assembly. Installation and testing & commissioning there is TT. The knowledge transfer is not much during construction due to the WPC's business policy. There is limitation. If we look at the TT, there is no new technology that is transferred. For example, the case of Boogie, it is one of their core business. So, there is no TT for that area. For Malaysian local companies, if we were to invest in the infrastructure for Boogie, the volume is not there. There is high requirement in the MRT project so it is not worth it for local companies to invest due to the small volumes.

The planning and requirement for TT is based on the ORD. MRT follows the guidelines in the ORD. If there is no ORD, MRT do not have any clear policy on TT.

Mechanism used is ORD and Offset programme. In MRT itself we try to leverage on the expatriate. What HR did is organize knowledge sharing session with the expatriate working in the project from MRT, ICE and PDP. Classroom and also working together with the expatriates.

In Offset, we follow the Offset programme. Internal process, with expatriates we plan the knowledge sharing once a month. In terms of monitoring, we have two times a year we evaluate our MRT staff recipients together with technology provider. We interview, record and give assignment. We also do rotation system to the staff.

In terms of communication channel, all the recipients report to WPC and WPC's JV. We have also monthly meeting with WPCs. The communication channel is moderate. Improvement that can be done, the problem is they put under JV. It should be put direct under the WPC. The recipients receive two instructions from WPC and WPC's JV partner. Only one side monitor the recipients.

Modes of TT – OJT, Knowledge sharing, process development, test equipment, testing, Modes most used – OJT. Effectiveness – Human capital is OJT.

Criteria of result – Weakness of the programme is that it is not properly planned. The recipients are fresh graduates and only exposed to assembly work. It is good if they are exposed to operation and testing. They should attach them with other project not just at MRT project. They should focus on operation and testing. Result and competency is in specialized areas.

Measurement – Measure through the two times a year evaluation meeting. In terms of result, not necessary on knowledge, it is also on communication, working experience. Some of the programme such as the signalling package, the result is good. But not much with other WPC.

No innovation occurs from Technology transfer due to them only exposed to assembly process only.

Lessons learned, the programme should be plan properly. What is the objective and how to achieve the objective must be clear. The plan for them to be subject matter expert but that does not happen.

Who should the lead or driver for the TT to achieved the objectives? The IMU should take the lead because the ORD is prepared by them so they must make sure the programme achieve the objectives set in the ORD. Each parties need to be clear on what their role. Objectives must be clear and in line with the Job OJT. There is misalignment between the objectives and the job OJT set to the recipients. We prefer placement with the OEMs. Another lessons learned, the recipients should have one or two years' experience. So they would know what knowledge to absorb and what questions to ask. The transfer did not happen because the recipients did not know what to ask. They even do not know the technology is new or not. If the WPC is cooperative, the recipient should also be from people that have more experience for more effective technology transfer.

Respondent A6

There is no requirement of TT in MRT. Solely depending on ORD. Current mechanism for TT is Offset programme. Besides offset, not much TT programme.

We transfer staff from line 1 to line, 2. We also have lessons learned workshop. We shared what is good and what is bad during the workshop. So, there is knowledge transfer happened.

Project team do the planning for the lessons learned workshop and the TT programme. In terms of monitoring, we come up with the outcome of the lessons learned we put it in the tender requirement for line, 2. Monitoring is done through project management such as project meeting.

From communication, technical requirement comes from project but if it concerns policy it relates back to procurement and HR.

Any improvement to the communication, depend on the personnel in the project team. If the personnel is very active and interested in the offset programme, there should not be any problem in terms of the communication.

Procurement just playing coordinating role and liaising between TDA and project team. Procurement should be there but the project team should be leading in terms of developing the TT programme. Procurement as coordinating role is effective currently. MRT channel for TT are Technology Transfer programme, knowledge sharing – classroom type, once a month.

In terms of knowledge transfer of WPC to MRT is more to OJT and on-site learning. Main mode used are OJT and on-site learning. The most effective TT are OJT and on-site learning. Knowledge sharing (Lessons Learned) is effective or not? We can know what happen in Line 1. We try to avoid in happening in Line, 2.

TT criteria for a result can be improvement of knowledge from our MRT staff. Increase of competency. Measurement is more on job performance on whether they can manage the suppliers better. For example, the trainee from Offset they are better performing in the job. Job performance appraisal is used to measure. There is also increase of salary and promotion.

In terms of innovation, not much I can see. Lessons learned, the technology transfer should focus more on the design and interface. Not much on construction. Improvement we can get more people to be involved in the design. Most Malaysian involved in construction but not much during the design phase. For track work, it should focus on design not much on construction.



Appendix C TT Models from the Literature Review

Conceptual model for international TT in construction projects (Waroonkun, 2007)



Influence Factors of Technology Transfer (Lai and Chao, 2006)



TT Conceptual Framework (A. Khan, 2011)



Amalgamated Model of International Technology Transfer Project Models in Malaysia (Omar *et al.*, 2008)



Capacity and Capability Development in Indigenous Construction Firms through Technology Transfer in Construction: A Malaysia Experience (A. Bakar, 2006)



Technology Transfer in the Construction Industry (Uusitalo and Lavikka, 2020)



Modelling a Conceptual Framework of Technology Transfer Process in Construction Projects: An Empirical Approach (Owusu-Manu *et al.*, 2017)



Conceptual TT Model for managing technology transfer for public R&D institutions in India (Purushotham *et al., 20*15)



Technology Transfer Model Process: Planning and Implementation Stage (Kundu *et al., 20*15)



An Empirical Study on Factors Influencing Technology Transfer Using Structural Equation Modelling (Alizadeh *et al.*, 2018)

Appendix D External Expert Verification of Question Interviews

INDUCTION INTERVIEW QUESTIONS

Induction Interview Questions	Expert Reviewer's Comment
1. Conceptions of technology	
A. What are the agency's requirements for	
Technology Transfer?	
B. How does the agency identify the requirements	
for Technology Transfer?	
C. What are the current Technology Transfer	
used?	
2. Technological Activity and transfer	
A.How do you or the agency do the planning for	
Technology Transfer Program/Projects?	
B. Do you put any proper mechanism to monitor	
the activities?	
3. Communication channels	
A. What are the current communication channels	
for Technology Transfer process and activities?	
D Doog the communication channels offective?	
B.Does the communication channels effective?	
C If not what are the improvement to the	
communication channels?	
communication channels.	
4. Modes of Technology Transfer?	
A. What are the Modes of Technology Transfer?	
(Classroom, OJT, Site Visit, Supervision)	
B. What are the most used modes of Technology	
Transfer?	
C. Which modes in your opinion is the most	
effective?	
5. Factors affecting Technology Transfer	
Does this factor affects? Yes or No	
Transfer Environment	
- Complexity of Technology	
- Construction Mode (e.g. JV vs Sub-	
contracting)	

- Government Policy	
- Government Enforcement	
Logrning Environment	
Palationship between WDCs and	
- Relationship between writes and Province	
Mutual Trust	
- Mutual Hust	
- Clear Understanding of the scope of	
Effective communication	
- Effective communication	
- Strong communent from senior	
Transferrent of wPCs and Recipients	
- Teamwork	
- Supervision	
wPCs Characteristics	
- Willingness to implement	
- Degree of Experience	
- Appropriate Management practices and	
approaches	
Recipient's Characteristics	
- Willingness to learn	
 Experience working with Foreigners 	
 Appropriate Management practices and 	
approaches	
- Adequate knowledge base	
6. Impact/Results of Technology Transfer	
A.What do you consider as the Impact/results of	
Technology Transfer in the project?	
(Competitiveness, financial, quality)	
B. What are the measurement used by the agency	
to measure Technology Transfer results?	
7. Gaps in the Technology Transfer Process	
A. What are the Lessons Learned from the	
previous technology transfer process?	
B. What are further improvement that you can	
suggest to technology transfer process?	

Name of Expert Reviewer:

[Name]

[Position]

[Organisation]

Signature :

Date :

Appendix E Survey Form (Primary Study)

QUESTIONNAIRE SURVEY

DEVELOPMENT OF A TECHNOLOGY TRANSFER MODEL FOR THE KLANG VALLEY MASS RAPID TRANSIT (KVMRT) PROJECT

This survey aims to gather data for measuring the effectiveness of Technology Transfer in the KVMRT Project. You are requested to kindly fill in the questionnaire. The information will be used for academic purposes only. Individual responses will be kept confidential. Only a consolidated summary of the result may be published. You may contact the researcher for any enquiries. Your cooperation is highly appreciated.

Definition of Technology Transfer

For the purpose of this Study, Technology Transfer (TT) can be defined as when all types of knowledge relating to the field (e.g. design, process, material use, equipment utilization, etc.) are transferred from a foreign party (transferor) to a host party (transferee) that arranges to receive it.

RESPONDENT'S BACKGROUND

Type of Organization:

Main Sub-co Contractor	ontractor Consu tant	ıl- 🗌 Unive	rsity Govern ment	- Others
Sex: Age:	e 🗌 Femal	е		
Below, 20	20 – 30 3	0 – 40	40 – 50	50+
Education:				
SPM Master Degree	DiplomaDoctorate	e Degree	Bachelor D	egree
How many years of working e	experience do you ha	ave?		
General 0 – 5	6 - 10	🗌 11 - 15	☐ 16 -, 20	20+
Rail Industry 0 - 5	6 - 10	11 - 15	☐ 16 -, 20	20+
How many projects have you	been involved that h	ave Technology	Transfer Program	me?
□ o	1 - 5	6 - 9		10+
How many <u>trainings or knowl</u>	edge transfer sessio	on have you part	cipated that relates	to a Technology

Transfer Programme?

1 - 5	6-9	10+

In your opinion, in which stage of construction does knowledge / technology transfer occurs the most?

Preliminary
Design

Detailed
Design

Construction / Manufacturing

Installation, Testing & Commissioning

In the following sections, please rate the degree of importance and relevance of this factor in your experience in the implementation of a Technology Transfer (TT) Program

PART A. ENABLERS: Please CIRCLE your selected rating number in COLUMN A

1	2	3	4	5
Very Low	Low	Moderate	High	Very High

Item Code	F1 – <u>TECHNOLOGY TRANSFER PLANNING</u> : concerned with the planning of the main project	COLUMN A - Rate the degree of importance and relevance of the factor				of factor
	with relation to Technology Transfer process	Very Low	Low	Mode- rate	High	Very High
T1	Initial Planning (industry gaps requirement, etc.) is critical to ensure a successful Technology Transfer program	1	2	3	4	5
Τ2	Preliminary assessment and selection of the potential technology providers and recipients is critical to ensure a successful Technology Transfer program	1	2	3	4	5
Т3	A detail <u>technology transfer requirement</u> <u>included in the tender</u> for a work package contract is critical for a successful technology transfer program	1	2	3	4	5

Item Code	F2 – <u>TRANSFER ENVIRONMENT</u> : concerned with the overall environment of the main project	COLUMN A - Rate the degree of importance and relevance of the factor				of factor
	with relation to Technology Transfer process	Very Low	Low	Mode- rate	High	Very High
Τ4	The technology transfer channel (i.e. joint venture, licensing, sub-contracting, etc.) will affect the effectiveness of a technology transfer program	1	2	3	4	5
T5	Government policy is essential for encouraging Technology Transfer program	1	2	3	4	5
Т6	Government enforcement of related policies is essential for encouraging Technology Transfer program	1	2	3	4	5
T7	Effective coordination & monitoring is critical for the success of a Technology Transfer program	1	2	3	4	5
Item Code	F3 – <u>LEARNING ENVIRONMENT</u> : concerned	CO	COLUMN A - Rate the degree of			
Code	with the relationship and communication between	importa	ance an	u i cievali		laciol

	the technology provider and recipient in the TT process	Very Low	Low	Mode- rate	High	Very High
Т8	The <u>different culture or nationality</u> between the technology provider and recipient could impact the technology transfer process	1	2	3	4	5
Т9	Mutual trust will help to create a good relationship between the provider and recipient	1	2	3	4	5
T10	<u>Clear understanding of technology transfer</u> <u>scope</u> for both the provider and recipient is essential for effective technology transfer	1	2	3	4	5
T11	Effective communication is essential for successful Technology Transfer	1	2	3	4	5
T12	A <u>strong commitment by the senior</u> <u>management</u> to Technology Transfer programmes is essential for their success.	1	2	3	4	5
T13	Good teamwork between the provider and recipient project team encourages Technology Transfer.	1	2	3	4	5
T14	The technology provider should provide an adequate facility to ensure that Technology Transfer occurs.	1	2	3	4	5
T15	A high degree of interest by both provider and recipient is essential for a good learning environment	1	2	3	4	5
T16	The technology provider should ensure <u>sufficient</u> and close supervision so that the Technology Transfer process actually occurs	1	2	3	4	5

Item	F4 – <u>TECHNOLOGY PROVIDER</u> CHARACTERISTICS: concerned with the	CC impor	DLUMN A tance and	- Rate th d relevan	e degre ce of th	e of e factor
Code	characteristics of the technology provider and how they impact on the TT process.	Very Low	Low	Mode- rate	High	Very High
T17	The technology provider's willingness to implement Technology Transfer initiatives and being cooperative is critical to the success of the TT programmes	1	2	3	4	5
T18	The technology provider's degree of <u>experience</u> working on international projects has an impact on the successfulness of TT programmes	1	2	3	4	5
T19	Technology provider's <u>management practices</u> <u>and procedures</u> (i.e. leadership style, project management etc.) are essential for effective TT	1	2	3	4	5
T20	<u>Technology provider</u> with an <u>extensive</u> <u>knowledge base</u> in advanced project management and technology are essential for effective TT to occur.	1	2	3	4	5

T21	Technology provider need to be honest and	1	2	3	4	5
	transparent in their approach of the Technology					
	Transfer process					

Item	F5 – <u>RECIPIENT'S CHARACTERISTICS</u> :	CC	DLUMN	A - Rate t	he degre	e of
Code	Recipient of technology and how they impact on the Technology Transfer (TT) process.	Very Low	Low	nd releva Mode- rate	nce of th High	e factor Very High
T22	The recipient's <u>willingness to learn</u> new technologies from the technology provider and <u>change their existing work practices</u> has a bearing on the success of TT programmes	1	2	3	4	5
T23	The recipient's <u>degree of experience working</u> <u>with foreigner/technology provider</u> has an impact on the successfulness of TT programmes	1	2	3	4	5
T24	The recipient has <u>appropriate management</u> <u>practices and approaches</u> (i.e. leadership style, project management etc.) are essential for effective TT	1	2	3	4	5
T25	Recipient's with an <u>adequate knowledge base</u> to learn more advanced skills (i.e. sufficient formal training, qualifications and experience, etc.) are essential for effective TT to occur.	1	2	3	4	5

PART B. OUTCOMES: Please CIRCLE your selected rating number in COLUMN A

COLUMN A: From your experience, rate the actual outcome of these factors in your previous Technology Transfer (TT) project or exercise using the following scale below.

1	2	3	4	5
Very Low	Low	Moderate	High	Very High

ltem Code	O1 – <u>ECONOMIC ADVANCEMENT</u>: concerned with the improvement in economic conditions of	COL	JMN A - ac	Rate the	degree ome	of the
	your industry due to TT programmes.	Very Low	Low	Mode- rate	High	Very High
T26	TT programmes enhanced our company's competitiveness in the market.	1	2	3	4	5
T27	TT programmes enhanced our company's overall performance over the long-term (i.e. better management skills, improved quality, etc.)	1	2	3	4	5
T28	TT programmes enhanced our company's overall profitability over the long-term	1	2	3	4	5
T29	Technology transfer programmes allows our company to be more innovative and produce new products or services	1	2	3	4	5

Item Code	O2 – <u>KNOWLEDGE ADVANCEMENT</u>: concerned with the impact of TT programmes on	COLUMN A - Rate the degree of the actual outcome						
	the recipient's knowledge base (i.e. skills, practices, etc.)	Very Low	Low	Mode- rate	High	Very High		
Т30	TT programmes help improved the <u>knowledge</u> <u>and skill</u> of local workers.	1	2	3	4	5		
T31	Recipients who have gained new knowledge from TT programmes utilize this knowledge to enhance their working practices.	1	2	3	4	5		
T32	Technology transfer enhances local workers <u>competency</u> and accelerate their career progression and increase their salary over the long term.	1	2	3	4	5		

Item Code	O3 – <u>PROJECT PERFORMANCE</u> : concerned with the impact of TT programmes on the	COLUMN A - Rate the degree of the actual outcome						
	performance of design and projects.	Very Low	Low	Mode- rate	High	Very High		
Т33	TT programmes enhanced the project financial <u>performance</u> (i.e. project budget, profitability, cash flow, etc.) of our projects.	1	2	3	4	5		
Т34	TT programmes enhanced the project schedule performance (i.e. on-time project completion, resource management, etc.) of our projects.	1	2	3	4	5		
T35	TT programmes enhanced the quality <u>standards</u> on our projects (i.e. less re-work, satisfied clients and improved products).	1	2	3	4	5		

Background Information (*Optional)

Could you please provide some information about yourself?

Name: _____
Position: _____

Name of Organization: _____

Contact Address: _____

Phone: ______ E-Mail: _____

Appendix F Then	e Frequencies from	NVivo 12 A	Analysis ((Primary	Study 1	1)
-----------------	--------------------	------------	------------	----------	---------	----

Theme	Frequency	Meaning	Evidence
Technology	10	Any activities	"For MRT
Transfer		concerning the initial	requirement,
Planning		preparation of the	technology transfer is
		implementation for	related to the technical
		the technology	team, so we requested
		transfer process and	respective Systems
		programme (Rose,	team what their
		1995; Smith, 1995).	requirements [sic.]."
			(A2)
			"The objective of the
			technology transfer is
			defined together with
			Malaysian Industry
			Government Group for
			High Technology
			(MIGHT) at that time.
			MIGHT already have
			the gap analysis and
			rail blueprint." (A3)
Technology	13	The channel for the	"Joint Venture is much
Transfer Channel		technology to flow	more effective as they
		from its provider to	will do it together and
		the recipient (A.	that is more effective."
		Khan, 2011;	(A3)
		Waroonkun, 2007).	

Theme Frequencies 1	Identified	from NVivo	12 Analy	/sis
---------------------	------------	------------	----------	------

Theme	Frequency	Meaning	Evidence
			"There is technology
			transfer in the process
			of assembly,
			installation, testing
			and commissioning."
			(A6)
Learning	16	The soft skill or hard	"On-Job-Training
Environment		infrastructure put in	(OJT) is the best, but
		place for technology	you cannot
		transfer (Awang et	accommodate all.
		al., 2008; Jusoff et	There is a limitation to
		al., 2009).	participants." (A1)
			"During OJT they get
			to get to be involved in
			the process even
			though at that time
			they do not know the
			right [sic.]." (A3)
Government	4	The specific	"MRT follows the
Policy		government policy in	Offset Requirement
		encouraging and also	Document (ORD) set
		enforcing technology	by the government for
		transfer in the project	the technology
		(Balakrishnan, 2007;	transfer." (A1)
		Dinakar, 2011).	
			"The Offset
			Requirement
			document given by the
			government is only for
			the tender stage." (A2)

Theme	Frequency	Meaning	Evidence
Technology	8	The overall	"The government
Transfer		ecosystem governing	policy is there only to
Environment		the KVMRT	make sure the Works
		technology transfer	Package Contractors
		programme from	fulfil their Offset
		planning to	obligations." (A4)
		implementation (A.	"The government
		Khan, 2011; Awang	already have the gap
		<i>et al.</i> , 2008).	analysis and
			blueprint." (A4)
Coordination &	8	The coordination and	"Because of offset, we
Monitoring		monitoring of the	have meeting and
		overall technology	progress update [sic.].
		transfer programme	There is a structured
		from planning to	approach." (A2)
		implementation	
		(Hamdan, 2015;	"Monitoring is done
		Rose, 1995;	through project
		Waroonkun, 2007).	management such as
			project meetings."
			(A6)
Communication	6	The communication	"The communication
Channel		flow between related	mechanism is
		parties and	effective. Will
		stakeholders with	improve better with no
		regards to the	intervention from
		technology transfer	external parties." (A1)
		programme (Khayat,	
		2015; Waroonkun,	"The communication
		2007).	channel is good
			because we have the
			MRT ICP Committee

Theme	Frequency	Meaning	Evidence
			and the working
			committee." (A3)
Technology	6	The technology	"Due to the technology
Provider's		provider's	[being] the WPC's
Characteristics		organisational	core business, there is
		behaviour and	limited technology
		structure that relates	transfer." (A5)
		to implementing the	
		technology transfer	"We need to consider
		programme (Khayat,	WPC's limitation such
		2015; Rose, 1995).	as space, trainer
			availability and
			facilities." (A3)
Technology	13	The technology	"The recipient must be
Recipient's		recipient's	ready. They should not
Characteristics		organisational	use this opportunity as
		behaviour and	a way to get jobs."
		structure that relates	(A4)
		to implementing the	
		technology transfer	"The recipient should
		programme (A.	be from people who
		Khan, 2011;	have experience [of]
		Waroonkun, 2007).	an effective
			technology transfer."
			(A5)

Appendix G Pearson Correlation Matrix

Pearson Correlation Matrix (T1 to T17)

	T1	T2	T3	T4	T5	T6	T7	T8	Т9	T10	T11	T12	T13	T14	T15	T16	T17
T1	1																
T2	0.743 **	1															
Т3	0.512 **	0.610 **	1														
T4	0.347 **	0.405 **	0.443 **	1													
Т5	0.372 **	0.387 **	0.431 **	0.443 **	1												
T6	0.352 **	0.361 **	0.400 **	0.308 **	0.782 **	1											
T7	0.441 **	0.443 **	0.469 **	0.402 **	0.588 **	0.618 **	1										
T8	0.053	0.091	0.045	0.172 *	0.014	-0.056	0.062	1									
Т9	0.317 **	0.362 **	0.356 **	0.254 **	0.284 **	0.224 **	0.389 **	0.235 **	1								
T10	0.426	0.421 **	0.430	0.312 **	0.461 **	0.446 **	0.551 **	0.073	0.478 **	1							
T11	0.398 **	0.341 **	0.380 **	0.259 **	0.406 **	0.382	0.497 **	0.118	0.447 **	0.691 **	1						

	T1	T2	T3	T4	T5	T6	T7	T8	Т9	T10	T11	T12	T13	T14	T15	T16	T17
T12	0.370 **	0.384 **	0.390 **	0.379 **	0.437 **	0.425 **	0.548 **	0.113	0.419 **	0.584 **	0.629 **	1					
T13	0.371 **	0.385 **	0.392 **	0.307 **	0.371 **	0.323 **	0.504 **	0.155 *	0.508 **	0.587 **	0.634 **	0.627 **	1				
T14	0.285 **	0.352 **	0.391 **	0.265 **	0.337 **	0.256 **	0.454 **	0.157 *	0.354 **	0.474 **	0.498 **	0.542 **	0.487 **	1			
T15	0.337 **	0.359 **	0.421 **	0.296 **	0.411 **	0.403 **	0.481 **	0.060	0.389 **	0.571 **	0.543 **	0.574 **	0.554 **	0.519 **	1		
T16	0.289 **	0.330 **	0.336 **	0.359 **	0.320 **	0.274 **	0.434 **	0.150 *	0.409 **	0.488 **	0.523 **	0.500 **	0.555 **	0.547 **	0.598 **	1	
T17	0.323 **	0.355 **	0.407 **	0.283 **	0.409 **	0.375 **	0.452 **	0.068	0.368 **	0.551 **	0.542 **	0.482 **	0.501 **	0.472 **	0.536 **	0.532 **	1
T18	0.410 **	0.364 **	0.393 **	0.266 **	0.437 **	0.370 **	0.499 **	0.152 *	0.470 **	0.530 **	0.539 **	0.455 **	0.417 **	0.507 **	0.536 **	0.543 **	0.659 **
T19	0.350 **	0.398 **	0.404 **	0.317 **	0.286 **	0.252 **	0.384 **	0.264 **	0.455 **	0.394 **	0.423 **	0.374 **	0.496 **	0.378 **	0.446 **	0.491 **	0.523 **
T20	0.438 **	0.402 **	0.429 **	0.336 **	0.366 **	0.325 **	0.397 **	0.106	0.447 **	0.484 **	0.484 **	0.390 **	0.465 **	0.403 **	0.448 **	0.440 **	0.590 **
T21	0.379 **	0.393 **	0.355 **	0.297 **	0.353 **	0.303 **	0.448 **	0.105	0.429 **	0.523 **	0.518 **	0.414 **	0.519 **	0.459 **	0.454 **	0.569 **	0.570 **
T22	0.388 **	0.359 **	0.465 **	0.272 **	0.450 **	0.360 **	0.457 **	0.030	0.433 **	0.564 **	0.553 **	0.465 **	0.490 **	0.352 **	0.507 **	0.429 **	0.538 **
T23	0.197 **	0.286 **	0.335 **	0.261 **	0.144 *	0.192 **	0.225 **	0.266 **	0.326 **	0.188 **	0.232 **	0.257 **	0.312 **	0.198 **	0.269 **	0.266 **	0.295 **

	T1	T2	T3	T4	T5	T6	T7	T8	Т9	T10	T11	T12	T13	T14	T15	T16	T17
T24	0.262 **	0.332 **	0.337 **	0.181 **	0.251 **	0.274 **	0.339 **	0.048	0.373 **	0.397 **	0.393 **	0.368 **	0.457 **	0.303 **	0.444 **	0.402 **	0.286 **
T25	0.209 **	0.338 **	0.304 **	0.303 **	0.253 **	0.247 **	0.368 **	0.145 *	0.407 **	0.381 **	0.353 **	0.393 **	0.444 **	0.375 **	0.345 **	0.438 **	0.403 **
T26	0.334 **	0.346 **	0.389 **	0.306 **	0.358 **	0.337 **	0.405 **	0.072	0.394 **	0.433 **	0.386 **	0.349 **	0.447 **	0.257 **	0.369 **	0.324 **	0.369 **
T27	0.303 **	0.355 **	0.412 **	0.287 **	0.335 **	0.272 **	0.324 **	-0.013	0.420 **	0.457 **	0.393 **	0.387 **	0.432 **	0.277 **	0.359 **	0.316 **	0.373 **
T28	0.305 **	0.300 **	0.396 **	0.161 **	0.278 **	0.174 *	0.160 *	0.025	0.302 **	0.270 **	0.287 **	0.203 **	0.331 **	0.241 **	0.305 **	0.268 **	0.379 **
T29	0.356 **	0.434 **	0.468 **	0.312 **	0.428 **	0.394 **	0.373 **	0.060	0.390 **	0.468 **	0.414 **	0.333 **	0.438 **	0.298 **	0.441 **	0.386 **	0.437 **
T30	0.353 **	0.371 **	0.463 **	0.290 **	0.385 **	0.346 **	0.424 **	-0.053	0.294 **	0.379 **	0.301 **	0.361 **	0.385 **	0.369 **	0.445 **	0.336 **	0.384 **
T31	0.399 **	0.435 **	0.426 **	0.289 **	0.412 **	0.340 **	0.382 **	0.051	0.354 **	0.392 **	0.422 **	0.297 **	0.473 **	0.335 **	0.461 **	0.409 **	0.435 **
T32	0.405 **	0.319 **	0.333 **	0.230 **	0.333 **	0.289 **	0.401 **	0.041	0.306 **	0.390 **	0.406 **	0.343 **	0.416 **	0.311 88	0.419 **	0.395 **	0.402 **
Т33	0.287 **	0.324 **	0.324 **	0.220 **	0.223 **	0.169 *	0.230 **	0.138	0.346 **	0.348 **	0.254 **	0.248 **	0.421 **	0.157 *	0.336 **	0.246 **	0.283 **
T34	0.257 **	0.294 **	0.296 **	0.174 *	0.229 **	0.224 **	0.305 **	0.100	0.395 **	0.382 **	0.321 **	0.254 **	0.353 **	0.157 *	0.277 **	0.260 **	0.292 **
T35	0.444 **	0.395 **	0.384 **	0.224 **	0.277 **	0.255 **	0.349 **	0.120	0.298 **	0.465 **	0.435 **	0.365 **	0.397 **	0.198 **	0.391 **	0.330 **	0.361 **

* Correlation is significant at the 0.05 level (2-tailed)

** Correlation is significant at the 0.01 level (2-tailed)

	T18	T19	T20	T21	T22	T23	T24	T25	T26	T27	T28	T29	T30	T31	T32	T33	T34	T35
T19	0.569 **	1																
T20	0.613 **	0.607 **	1															
T21	0.572 **	0.577 **	0.640 **	1														
T22	0.481 **	0.374 **	0.491 **	0.486 **	1													
T23	0.302 **	0.361 **	0.275 **	0.276 **	0.298 **	1												
T24	0.370 **	0.345 **	0.310 **	0.270 **	0.431 **	0.474 **	1											
T25	0.361 **	0.380 **	0.323 **	0.375 **	0.402 **	0.518 **	0.478 **	1										
T26	0.300 **	0.411 **	0.384 **	0.446 **	0.456 **	0.291 **	0.389 **	0.386 **	1									
T27	0.328 **	0.413 **	0.390 **	0.378 **	0.458 **	0.255 **	0.340 **	0.342 **	0.711 **	1								
T28	0.282 **	0.381 **	0.339 **	0.299 **	0.399 **	0.260 **	0.310 **	0.179 *	0.584 **	0.614 **	1							

Pearson Correlation Matrix (T18 to T35)

	T18	T19	T20	T21	T22	T23	T24	T25	T26	T27	T28	T29	T30	T31	T32	T33	T34	T35
T29	0.373	0.405	0.387	0.412	0.484	0.374	0.422	0.308	0.585	0.640	0.558	1						
	**	**	**	**	**	**	**	**	**	**	**							
T30	0.387	0.347	0.371	0.364	0.499	0.305	0.340	0.305	0.453	0.524	0.469	0.531	1					
	**	**	**	**	**	**	**	**	**	**	**	**						
T31	0.477	0.452	0.426	0.429	0.528	0.375	0.374	0.389	0.446	0.493	0.487	0.582	0.576	1				
	**	**	**	**	**	**	**	**	**	**	**	**	**					
T32	0.474	0.373	0.429	0.509	0.465	0.310	0.325	0.356	0.490	0.463	0.397	0.508	0.637	0.620	1			
	**	**	**	**	**	**	**	**	**	**	**	**	**	**				
T33	0.269	0.331	0.272	0.266	0.281	0.402	0.401	0.304	0.478	0.386	0.407	0.422	0.372	0.400	0.371	1		
	**	**	**	**	**	**	**	**	**	**	**	**	**	**	**			
T34	0.325	0.407	0.319	0.274	0.266	0.392	0.369	0.283	0.417	0.414	0.320	0.466	0.359	0.360	0.388	0.688	1	
	**	**	**	**	**	**	**	**	**	**	**	**	**		**	**		
T35	0.386	0.309	0.356	0.396	0.411	0.317	0.321	0.251	0.454	0.443	0.412	0.520	0.522	0.427	0.540	0.512	0.558	1
	**	**	**	**	**	**	**	**	**	**	**	**	**	**	**	**	**	

* Correlation is significant at the 0.05 level (2-tailed)

** Correlation is significant at the 0.01 level (2-tailed)

LIST OF PUBLICATIONS

Journal with Impact Factor

- Hamdan, A. R., Fathi, M. S. & Mohamed, Z. (2018). Evolution of Malaysia's Technology Transfer Model Facilitated by National Policies, International Journal of Engineering & Technology, 7(2.29), 196-202. (Indexed by SCOPUS)
- Hamdan, A. R., Fathi, M. S. & Mohamed, Z. (2018a). The Critical Success Factors for the Technology Transfer in Klang Valley Mass Rapid Transit Project, The Journal of Social Sciences Research(2), 166-172. (Indexed by SCOPUS)
- 3. Hamdan, A. R., Fathi, M. S. and Mohamed, Z. (2020). Leveraging Public Infrastructure Project as a Driver for Technology Development - A Case Study on a Technology Transfer Model for the Klang Valley Mass Rapid Transit Development Project, International Journal of Research and Innovation in Social Science, IV(VIII), 408 -420. (IF (2014):5.611)

Non-Indexed Conference Proceedings

 Hamdan, A. R., Fathi, M. S. & Mohamed, Z. (2017). Technology Transfer in the Klang Valley Mass Rapid Transit Project: Key Success Factors, Paper presented at 1st International Conference on Sustainable Infrastructure and Engineering 2017. Kuala Lumpur.